

MSATS PROCEDURES: PROCEDURE FOR THE MANAGEMENT OF WHOLESALE, INTERCONNECTOR, GENERATOR AND SAMPLE (WIGS) NMIS

PPEPARED FOR: Electricity

PREPARED BY: Retail Markets and Metering

DOCUMENT NO: MT_RT1700v004.10

VERSION NO: 4.01

EFFECTIVE DATE: 15 May 2014 1 July 2014

FINAL DETERMINATION INITIAL CONSULTATION FINAL DETERMINATION

Australian Energy Market Operator Ltd ABN 94 072 010 327

www.aemo.com.au info@aemo.com.au

TASMANIA

NEW SOUTH WALES QUEENSLAND SOUTH AUSTRALIA VICTORIA AUSTRALIAN CAPITAL TERRITORY

QUICK REFERENCE GUIDE

Change FRMP

Change Reason Code	Description	Chapter & Page No
1000, 1020	Change FRMP	Ch 2, Page 13
1050, 1051	Change FRMP - Where FRMP is not the RP	Ch 3, Page 18

Provide Data

Change Reason Code	Description	Chapter & Page No
1500	Submit Actual Change Date	Ch 4, Page 22

Create NMI

Change Reason Code	Description	Chapter & Page No
2000, 2001	Create NMI	Ch 5, Page 24
2500, 2501	Create NMI, Meter, Datastream	Ch 6, Page 27

Maintain Profile Shapes

Change Reason Code	Description	Chapter & Page No
2100, 2101	Create External Profile Shape	Ch 7, Page 31
5110, 5111	Change External Profile Shape	

Maintain Metering

Change Reason Code	Description	Chapter & Page No
3000, 3001	Create Metering Details	Ch 8, Page 34
3004, 3005	Exchange of Metering Information	Ch 9, Page 37
3050, 3051	Change Metering Details	Ch 10, Page 40
3100, 3101	Change Network Tariff Code	Ch 11, Page 43

Maintain Datastream

Change Reason Code	Description	Chapter & Page No
4000, 4001	Create Datastream Details	Ch 12, Page 45
4004, 4005	Exchange of Datastream Information	Ch 13, Page 47
4050, 4051	Change Datastream Details	Ch 14, Page 49

Maintain NMI

Change Reason Code	Description	Chapter & Page No
5001	Backdate NMI Start Date	Ch 15, Page 51
5050, 5051	Change NMI	Ch 16, Page 54

Change Roles (excluding FRMP)

Change Reason Code	Description	Chapter & Page No
6100, 6110	Change LNSP	Ch 17, Page 57
6200, 6210	Change MDP	Ch 18 Page 59
6300, 6301	Change RP	Ch 19, Page 62
6400, 6401	Change LR	Ch 20, Page 65
6421	Change LR Embedded Network (Child) Generator	Ch 21, Page 68
6500, 6501	Change ROLR	Ch 22, Page 70
6700, 6701	Change MP	Ch 23, Page 72

AEMO only

Change Reason Code	Description	Chapter & Page No
5100, 5101	Standing data updates	Ch 24, Page 74
BCxx	Invoke Bulk Change Tool	Ch 25, Page 78

Please See DISCLAIMER on the Last Page of this document.

HOW TO USE THIS DOCUMENT

The MSATS Procedures: Procedures for the Management of Wholesale, Interconnector, Generator and Sample (WIGS) NMIs (as amended time to time) is a subset of the MSATS Procedures: CATS Procedures Principles and Obligations (as amended time to time).

This document is commonly known as the WIGS Procedures. The WIGS Procedures contain the principles that govern consumer transfer, the registration of metering installation and management of standing data for NMIs that are classified as wholesale, interconnector, generator or sample.

The WIGS Procedures must be read in conjunction with the *MSATS Procedures: CATS Procedures Part 1 Principles and Obligations*. Details on general obligations by role, CATS functionality and transaction type, CATS codes and rules for a change request and Access to CATS Standing data are explained in Chapters 1 – 5 & 44 of the *MSATS Procedures: CATS Procedures Principles and Obligations*. However specific obligations by roles are detailed in chapters within this document.

If you receive a notification from another participant for a NMI with a NMI classification code that is covered by these WIGS Procedures and are not sure what obligations you have, confirm, from the notification, the change reason code. Then check the 'Quick Reference Guide' at the beginning of this document to confirm what chapter covers this change reason code.

The revision of these procedures is detailed in Chapter 26.

CONTENTS

QUI	CK REF	ERENCE GUIDE	. 2
НΟ	N TO U	SE THIS DOCUMENT	. 3
1.	INTRO	DUCTION	. 9
	1.1 1.2 1.3 1.4 1.5 1.6 1.7 1.8 1.9 1.10 1.11	CONTEXT FOR THE WIGS PROCEDURES PURPOSE AND APPLICATION OF WIGS PROCEDURES. AMENDMENT TO THESE PROCEDURES ACCESS TO DATA DISPUTE RESOLUTION ENFORCEABILITY OF THE PROCEDURES COUNTING OF DAYS. EFFECTIVE TIME OF CHANGE RELATED DOCUMENTS. JURISDICTIONS NMI CLASSIFICATION CODES PROCEDURE DEVELOPMENT HISTORY	10 10 10 10 10 10 11
2.		SE FRMP - CHANGE FRMP	
	2.1 2.2 2.3 2.4 2.5 2.6 2.7 2.8 2.9 2.10 2.11	APPLICATION [1000 1020] CONDITIONS PRECEDENT INITIATION RULES FRMP OBLIGATIONS LNSP OBLIGATIONS MDP OBLIGATIONS RP OBLIGATIONS MPB OBLIGATIONS TIME FRAME RULES OBJECTION RULES CHANGE REQUEST STATUS NOTIFICATION RULES	13 13 14 14 15 15
3.	CHANG	SE FRMP – WHERE FRMP IS NOT THE RP	18
	3.1 3.2 3.3 3.4 3.5 3.6 3.7 3.8 3.9 3.10 3.11	APPLICATION [1050 1051] CONDITIONS PRECEDENT INITIATION RULES FRMP OBLIGATIONS LNSP OBLIGATIONS MDP OBLIGATIONS RP OBLIGATIONS TIME FRAME RULES OBJECTION RULES CHANGE REQUEST STATUS NOTIFICATION RULES	18 18 19 19 20 20 20
4.	PROVI	DE DATA – PROVIDE ACTUAL CHANGE DATE	22
	4.1 4.2 4.3 4.4 4.5 4.6	APPLICATION [1500] INITIATION RULES MDP OBLIGATIONS. TIME FRAME RULES OBJECTION RULES CHANGE REQUEST STATUS NOTIFICATION RULES	22 22 22 22 23
5.	CREAT	E NMI - CREATE A NMI	
	5.1 5.2 5.3 5.4 5.5	APPLICATION [2000 2001] CONDITIONS PRECEDENT INITIATION RULES LNSP OBLIGATIONS MPB OBLIGATIONS	24 24 24

	5.6 5.7	TIME FRAME RULES	
	5.8	CHANGE REQUEST STATUS NOTIFICATION RULES	
6.	CREAT DATAS	E NMI – CREATE NMI, METERING INSTALLATION DETAILS AND MDM	27
	6.1	Application [2500 2501]	27
	6.2	CONDITIONS PRECEDENT.	
	6.3	INITIATION RULES	
	6.4	LNSP OBLIGATIONS	
	6.5	TIME FRAME RULES	
	6.6	OBJECTION RULES	
7.	6.7 MAINT	CHANGE REQUEST STATUS NOTIFICATION RULES	30
•			31
	7.1	APPLICATION [2100 2101 5110 5111]	
	7.2	CONDITIONS PRECEDENT	
	7.3	INITIATION RULES	
	7.4 7.5	AEMO OBLIGATIONS	
	7.5 7.6	TIME FRAME RULES OBJECTION RULES	
	7.0 7.7	CHANGE REQUEST STATUS NOTIFICATION RULES	
8.		AIN METERING – CREATE METERING INSTALLATION DETAILS	
	8.1	APPLICATION [3000 3001]	
	8.2	CONDITIONS PRECEDENT	
	8.3	INITIATION RULES	
	8.4	MPB OBLIGATIONS	
	8.5 8.6	TIME FRAME RULES OBJECTION RULES	
	8.7	CHANGE REQUEST STATUS NOTIFICATION RULES	
9.	_	AIN METERING - EXCHANGE OF METERING INFORMATION	
	9.1	APPLICATION [3004 3005]	
	9.2	CONDITIONS PRECEDENT.	
	9.3	INITIATION RULES	-
	9.4	MPB OBLIGATIONS	_
	9.5	TIME FRAME RULES	00
	9.6	OBJECTION RULES	
	9.7	CHANGE REQUEST STATUS NOTIFICATION RULES	
10.		ANN METERING - CHANGE METERING INSTALLATION DETAILS	
	10.1 10.2	APPLICATION [3050 3051]	
	10.2	INITIATION RULES	
	10.4	MPB OBLIGATIONS	
	10.5	TIME FRAME RULES	
	10.6	OBJECTION RULES	
	10.7	CHANGE REQUEST STATUS NOTIFICATION RULES	
11.		AIN METERING – CHANGE NETWORK TARIFF CODE	
	11.1	APPLICATION [3100 3101]	
	11.2	CONDITIONS PRECEDENT	
	11.3	INITIATION RULES	
	11.4 11.5	LNSP OBLIGATIONS TIME FRAME RULES	
	11.6	OBJECTION RULES	
	11.7	CHANGE REQUEST STATUS NOTIFICATION RULES	
12.	MAINT	AIN DATASTREAM – CREATE MDM DATASTREAM	45
	12.1	APPLICATION [4000 4001]	45

	12.2	CONDITIONS PRECEDENT	45		
	12.3	INITIATION RULES	45		
	12.4	MDP OBLIGATIONS			
	12.5	TIME FRAME RULES			
	12.6	OBJECTION RULES			
	12.7	CHANGE REQUEST STATUS NOTIFICATION RULES	46		
13.	3. MAINTAIN DATASTREAM - EXCHANGE OF DATASTREAM INFORMATION				
	13.1	APPLICATION [4004 4005]	47		
	13.2	CONDITIONS PRECEDENT			
	13.3	INITIATION RULES	47		
	13.4	MDP OBLIGATIONS	47		
	13.5	TIME FRAME RULES	48		
	13.6	OBJECTION RULES	48		
	13.7	CHANGE REQUEST STATUS NOTIFICATION RULES	48		
14.	MAINT	AIN DATASTREAM - CHANGE MDM DATASTREAM	49		
	14.1	APPLICATION [4050 4051]	49		
	14.2	CONDITIONS PRECEDENT			
	14.3	INITIATION RULES			
	14.4	MDP OBLIGATIONS			
	14.5	TIME FRAME RULES			
	14.6	OBJECTION RULES			
	14.7	CHANGE REQUEST STATUS NOTIFICATION RULES			
15.	MAINT	AIN NMI – BACKDATE A NMI			
	15.1	APPLICATION [5001]			
	15.2	CONDITIONS PRECEDENT			
	15.3	INITIATION RULES			
	15.4	LNSP OBLIGATIONS			
	15.5	AEMO OBLIGATIONS	52		
	15.6	MPB OBLIGATIONS	52		
	15.7	TIME FRAME RULES	52		
	15.8	OBJECTION RULES			
	15.9	CHANGE REQUEST STATUS NOTIFICATION RULES	53		
16.	MAINT	AIN NMI – CHANGE A NMI	54		
	16.1	APPLICATION [5050 5051]	54		
	16.2	CONDITIONS PRECEDENT			
	16.3	INITIATION RULES			
	16.4	LNSP OBLIGATIONS	54		
	16.5	TIME FRAME RULES	55		
	16.6	OBJECTION RULES	55		
	16.7	CHANGE REQUEST STATUS NOTIFICATION RULES	55		
17.	CHAN	GE ROLE – CHANGE LNSP	57		
	17.1	APPLICATION [6100 6110]	57		
	17.2	CONDITIONS PRECEDENT			
	17.3	INITIATION RULES			
	17.4	LNSP OBLIGATIONS	57		
	17.5	TIME FRAME RULES			
	17.6	OBJECTION RULES			
	17.7	CHANGE REQUEST STATUS NOTIFICATION RULES			
18.	CHAN	GE ROLE – CHANGE MDP	59		
	18.1	APPLICATION [6200 6210]	59		
	18.2	CONDITIONS PRECEDENT			
	18.3	INITIATION RULES			
	18.4	FRMP OBLIGATIONS			
	18.5	RP OBLIGATIONS			
	18.6	MDP OBLIGATIONS			
	18.7	TIME FRAME RULES	60		

	18.8 18.9	OBJECTION RULES	
19.	CHAN	GE ROLE – CHANGE RP	62
	19.1	Application [6300 6301]	62
	19.2	CONDITIONS PRECEDENT	
	19.3	INITIATION RULES	
	19.4	RP OBLIGATIONS	
	19.5	FRMP OBLIGATIONS	
	19.6	TIME FRAME RULES	
	19.7	OBJECTION RULES CHANGE REQUEST STATUS NOTIFICATION RULES	
20	19.8	GE ROLE – CHANGE LR	
20.		APPLICATION [6400 6401]	
	20.1 20.2	CONDITIONS PRECEDENT	
	20.2	Initiation Rules	
	20.4	LR OBLIGATIONS	
	20.5	TIME FRAME RULES	
	20.6	OBJECTION RULES	
	20.7	CHANGE REQUEST STATUS NOTIFICATION RULES	
21.	CHAN	GE ROLE – CHANGE LR – EMBEDDED NETWORK (CHILD) – GENERATOR	67
	21.1	APPLICATION [6421]	
	21.2	CONDITIONS PRECEDENT	
	21.3	INITIATION RULES	
	21.4	LR OBLIGATIONS	
	21.5	TIME FRAME RULES	
	21.6 21.7	OBJECTION RULES CHANGE REQUEST STATUS NOTIFICATION RULES	
22			
22.		GE ROLE – CHANGE ROLR	
	22.1 22.2	APPLICATION [6500 6501] CONDITIONS PRECEDENT	
	22.2	Initiation Rules	
	22.4	ROLR OBLIGATIONS	
	22.5	TIME FRAME RULES	
	22.6	OBJECTION RULES	
	22.7	CHANGE REQUEST STATUS NOTIFICATION RULES	70
23.	CHAN	GE ROLE – CHANGE MPB AND / OR MPC	71
	23.1	Application [6700 6701]	
	23.2	CONDITIONS PRECEDENT	
	23.3	INITIATION RULES	
	23.4	RP OBLIGATIONS	
	23.5	MP OBLIGATIONS	
	23.6 23.7	LNSP OBLIGATIONSTime Frame Rules	
	23.8	OBJECTION RULES	
	23.9	CHANGE REQUEST STATUS NOTIFICATION RULES	
24.		ONLY - AEMO-INITIATED STANDING DATA UPDATES	
	24.1	APPLICATION [5100 5101]	
	24.1	CONDITIONS PRECEDENT	
	24.3	INITIATION RULES	
	24.4	AEMO OBLIGATIONS	
	24.5	TIME FRAME RULES	
	24.6	OBJECTION RULES	
	24.7	CHANGE REQUEST STATUS NOTIFICATION RULES	76
25.	AEMO	ONLY – CHANGE - ROLE, TNI OR DLF – BULK CHANGE TOOL (BCT)	77
	OF 1	INTRODUCTION	77

	25.2	CONDITIONS PRECEDENT	77
		AEMO OBLIGATIONS	
	25.4	REQUESTING PARTICIPANT OBLIGATIONS	78
	Note(1): THE REQUESTING PARTICIPANT MUST BE THE RP TO NOMINATE THE MDP, MPB AND M	РС
	•	ROLES	
	25.5	IMPACTED PARTICIPANT/S OBLIGATIONS	79
	25.6	MDP OBLIGATIONS	79
	25.7	TIME FRAME RULES	
	25.8	OBJECTION RULES	
	25.9	CHANGE REQUEST STATUS NOTIFICATION RULES	79
26.	WIGS F	PROCEDURES VERSION HISTORY	80
DIS	CLAIME	:R	82

1. INTRODUCTION

1.1 Context for the WIGS Procedures

- (a) The MSATS Procedures: Procedures for the Management of Wholesale, Interconnector, Generator and Sample NMIs ("WIGS Procedures") contain the principles that govern consumer transfer, the registration of metering installations and the management of standing data. The WIGS Procedures are an extension of the MSATS Procedures: CATS Procedures Principles and Obligations. These procedures are applicable to NMIs that have the classification of wholesale, interconnector, generator, sample, or external profiles. The procedures also identify the obligations placed on CATS participants.
- (b) MSATS is, in part, a system (the MSATS system) that is operated by AEMO to fulfil its obligations under the National Electricity Rules.
- (c) The Consumer Administration and Transfer Solution ("CATS") forms part of MSATS.

CATS is a market wide solution for:

- 1. The transfer of consumers between retailers where there is an implication for the wholesale settlements process for those retailers;
- The management of CATS standing data;
- 3. The administration of National Metering Identifier (NMI) registration; and
- 4. The facilitation of NMI discovery.
- (d) In the WIGS Procedures:
 - Words and phrases that appear in italics are defined in the glossary situated in the MSATS Procedures: CATS Procedures Principles and Obligations (as amended from time to time).
 - Words and phrases which are defined in the National Electricity Rules have the same meaning under the WIGS Procedures.

1.2 Purpose and Application of WIGS Procedures

1.2.1 Purpose

The purpose of the WIGS Procedures is to:

- (a) Define the attributes of a connection point for the purpose of consumer transfer.
- (b) Provide for the registration of a NMI and associated metering installation components against that connection point.
- (c) Facilitate market settlements.
- (d) Facilitate and support an efficient industry process for:
 - 1. The transfer of NMIs between retailers;
 - 2. The provision and maintenance of CATS standing data;
 - 3. The provision and maintenance of CATS codes and rules; and
 - 4. The discovery of approved NMI standing data.
- (e) Define the roles and obligations of CATS participants.
- (f) Enable Jurisdictions to convey to CATS participants the rules under which a retail transfer will be approved.

1.2.2 Application

- (a) The WIGS Procedures apply to Registered Participants in accordance with the Rules. They form part of the MSATS Procedures mentioned in Rules clause 7.2.8.
- (b) The procedures apply to Service Providers through the AEMO accreditation process that has been established in accordance with the Rules.
- (c) The WIGS Procedure document applies to NMI Class of Wholesale, Interconnector, Generator, External Profile Shape & Sample Meter. Refer to Chapters 1 to 5 and 44 of the

CATS procedure document for general chapters that relate to both CATS (Small & Large) and WIGS NMIs.

1.2.3 Regulatory Framework

- (a) The WIGS Procedures are subordinate to the National Electricity Rules.
- (b) The WIGS Procedures have been prepared in accordance with clause 7.2.8 of the National Electricity Rules.
- (c) A CATS Participant may also be bound by other relevant national and Jurisdictional provisions.

1.3 Amendment to these Procedures

- (a) The WIGS Procedures may be amended in accordance with clause 7.2.8(b) of the National Electricity Rules.
- (b) The WIGS Procedures shall be reviewed by AEMO as required.
- (c) Prior to introducing changes to the WIGS Procedures, AEMO shall facilitate a consultative process with Jurisdictions, Participants and Service Providers.
- (d) Changes to these procedures cannot be retrospective.
- (e) Approved changes to the WIGS Procedures will take effect at 00:00 hours Eastern Standard Time on the day of effect with due notice being given to Market Participants.

1.4 Access to Data

(a) A CATS Participant has a right to access data stored in CATS only as prescribed by the Standing Data Access Rules.

1.5 Dispute Resolution

- (a) Disputes which arise between Registered Participants in relation to the application or operation of this document must be resolved in accordance with the dispute resolution regime provided under the National Electricity Rules.
- (b) Disputes that arise between Service Providers and/or other CATS participants in relation to the application or operation of this document must be resolved in accordance with the dispute resolution regime contained in accreditation agreements established between these parties and AEMO.

1.6 Enforceability of the Procedures

(a) The WIGS Procedures will be enforceable in accordance with clauses 7.2.8(c), 7.2.8 (d) and 7.2.8 (e) of the National Electricity Rules.

1.7 Counting of Days

(a) Unless otherwise stated, any measure of days identified in these Procedures is given in business days. Business days are as defined in the National Electricity Rules.

1.8 Effective Time of Change

(a) All changes to NMI master records will become effective at 00:00 hours Eastern Standard Time on the day of effect. This includes consumer transfers and all changes to standing data.

1.9 Related Documents

The WIGS Procedures has been prepared with reference to the following documents:

- (a) National Electricity Rules.
- (b) MSATS Procedures: CATS Procedures Principles and Obligations (as amended from time to time).
- (c) NMI Procedures (AEMO document number ME_GN059v00x) (as amended from time to time).
- (d) MSATS User Reference Guide.

- (e) Metrology Procedure (Part A & B).
- (f) CATS Hints and Tips, which provides hints and tips on how to make best use of the CATS component of the MSATS software (as amended from time to time). This document may also be useful for Information Technology staff involved in developing systems that interface to MSATS.
- (g) Standing Data for MSATS (AEMO document number MT_MA1705v00x) (as amended from time to time).
- (h) Allocation of Embedded Network Codes (AEMO document number MT_GN1710v00x) (as amended from time to time).
- (i) Service Level Procedures for Metering Data Providers

1.10 Jurisdictions

(a) The jurisdictions that the WIGS Procedures cover are specified in Table 1-A.

Table 1-A - Jurisdiction codes

Code	Description
ACT	Australian Capital Territory
ALL	All Jurisdictions
NEM	National Electricity Market
NSW	New South Wales
QLD	Queensland
SA	South Australia
VIC	Victoria
TAS	Tasmania

1.11 NMI Classification Codes

- (a) The NMI classification code enables the MSATS system to be informed of the nature of the flow of electricity at the connection point to which the NMI information applies.
- (b) The NMI classification codes WHOLESAL, INTERCON, GENERATR, EPROFILE and SAMPLE are used by these procedures. They are parameters that can be used when defining change reason codes, application time frames and objection rules.
- (c) The NMI classification codes are based on the total annual load of the NMI as per Table 1-B.
- (d) The NMI classification codes 'WHOLESAL', 'INTERCON', 'GENERATR' and 'SAMPLE' relate to a NMI and not to a site.

The valid NMI classification codes are specified in Table 1-B.

Table 1-B - NMI classification codes

Code	Description	on ⁽²⁾
EPROFILE (1)	External P	rofile Shape
GENERATR ⁽¹⁾	Generator	
INTERCON ⁽¹⁾	Interconne	ector
LARGE	Victoria:	>=160 MWh
	NSW:	>=160 MWh
	ACT:	>= 160 MWh
	QLD:	>=100 MWh
	SA:	>=160 MWh
	TAS:	>=150 MWh
SAMPLE ⁽¹⁾	Sample M	eter
SMALL	Victoria:	<160 MWh
	NSW:	<160 MWh
	ACT:	< 160 MWh
	QLD:	< 100MWh
	SA:	<160 MWh
	TAS:	<150 MWh
WHOLESAL ⁽¹⁾	Wholesale	Transmission Node Identifier

Note (1): these codes are used in the WIGS Procedures.

Note (2): see relevant jurisdictional regulation for full details.

1.12 Procedure Development History

(a) These WIGS Procedures were developed following industry and jurisdictional consultation on the CATS Configuration rules, which determined the configuration that MSATS would utilise for NMIs with the classification of wholesale, interconnector, generator and sample. The document has been reviewed by the authorised Industry Reference and/or Working Groups and made available for public consultation.

2. CHANGE FRMP - CHANGE FRMP

2.1 Application [1000 1020]

This procedure applies to the following change reason codes:

1000 – Change FRMP

This is the situation where a new FRMP obtains the consent of the consumer to take responsibility for the supply of electricity to that connection point. The date of transfer would be a nominal prospective date. The NMI could transfer from a Tier-1 site to a Tier-2 site, or from a Tier-2 site to another Tier-2 site, or from a Tier-2 site to a Tier-1 site.

1020 – Change FRMP–Retro–Long Term/Error

This is the situation where a new FRMP obtains the consent of the consumer to take responsibility for the supply of electricity to that connection point. The date of transfer could be a date agreed by the current and new FRMP (a retrospective date) where the actual transfer date was in error. In addition, and for the Victorian jurisdiction only, the date of transfer could be a date agreed by the current and new FRMP (a retrospective date) to which both parties agree that the transfer should have been effected. The NMI could transfer from a Tier-1 site to a Tier-2 site, or from a Tier-2 site to another Tier-2 site, or from a Tier-2 site to a Tier-1 site.

2.2 Conditions Precedent

- (a) The NMI already exists in MSATS. If the NMI does not already exist in MSATS, the retailer should refer to Chapter 5 "Create a NMI" and seek LNSP action to establish the NMI.
- (b) The NMI classification is wholesale, interconnector, generator or sample.
- (c) Chapter 2 relates only to change reason codes 1000 and 1020.
- (d) The date of the transfer of retailer from the current FRMP to the new FRMP will be the actual change date.
- (e) The actual change date is to be established in accordance with the read type code submitted to the MSATS system by the new FRMP.

2.3 Initiation Rules

- (a) A new FRMP may initiate a change request to effect a change of retailer in accordance with clause 2.4.
- (b) The new FRMP must use one of the following change reason codes 1000 or 1020 to establish an initial change request.

2.4 FRMP Obligations

The new FRMP must:

- (a) Comply with the *consumer consent* provisions of the relevant jurisdiction for the NMI.
- (b) Obtain the NMI checksum from an approved source.
- (c) Confirm that the NMI is a valid NMI for the connection point prior to the initiation of a change request.
- (d) Submit a change request to MSATS to initiate the transfer of the consumer for which the new retailer has obtained an explicit informed consent.
- (e) Populate the initial change request with the following information:

Change reason code	Participant transaction ID	NMI and NMI checksum
Their participant ID	Proposed change date	New RP
Read type code		

(f) For change reason codes 1000 and 1020 where the new FRMP is the RP, and if the MDP is changing, populate the initial change request with the following additional information:

MDP		

(g) Nominate itself as the new FRMP.

The new FRMP may:

(h) Populate the initial change request with the identity of the following roles:

MPB	MPC	
-----	-----	--

or leave this action to the new RP to complete.

(i) For change reason code 1020 (retrospective), populate the initial change request with:

Actual end date	
71010.01.0110.010	

- (j) Nominate, in accordance with clause 2.4(e), a proposed change date, which is retrospective to the date on which the change request was initiated. The limits for retrospective transfer dates are provided in clause 2.9.
- (k) Withdraw a change of retailer transaction request at any time until the transfer is completed in MSATS.

The current FRMP may:

 Object to the change request in accordance with the objection rules provided in clause 2.10.

2.5 LNSP Obligations

The current LNSP must:

- (a) Provide the new MDP with the average daily load on request of that MDP.
- (b) Verify that the site is energised and if not facilitate the energisation of the site on the proposed transfer date or up to 2 days early. It is expected that requests for energisation will be made through a separate process between the FRMP and the LNSP to ensure that the transfer process does not delay the energisation of the site for the consumer.
- (c) If data is not already contained in the MSATS system, receive the data request. for the following data items:

LR	ROLR	TNI code
DLF code	NMI classification code	Jurisdiction code
NMI status code		

(d) Respond to a request from the MSATS system for data within 1 day of the date of the data request.

The current LNSP may:

(e) Object to the change request in accordance with the objection rules provided in clause 2.10.

2.6 MDP Obligations

The MDP must:

- (a) Once the transfer has completed, set up the NMI suffix(s) so that they become active on the actual change date.
- (b) On receipt of a data request for change reason code 1000, initiate a change request to provide the actual change date, which must be the date nominated as the proposed date by the new FRMP unless the date nominated is inappropriate, in which case the MDP must advise the new FRMP of the reason why the proposed date is inappropriate and mutually agree on what the actual change date should be.

- (c) Validate that the metering installation type code is identified as a type 1, type 2, type 3 or type 4 metering installation. If an error is identified, advise the new FRMP and await instructions.
- (d) For prospective changes that do not require a meter read, provide the actual change date within 2 days of the actual change date being reached.
- (e) For retrospective changes that do not require a meter read, provide the actual change date within 2 days of receipt of the data request.
- (f) For retrospective changes where the rules specify that an Actual Change Date is required, ensure that the actual change date submitted is no earlier than the time limit specified by the jurisdiction for the change reason code on the change request.
- (g) For example, if the jurisdictional time frame for a change reason code is 10 days, then the actual change date must be no earlier than 10 days prior to the date the data request was received, even if MSATS would allow an earlier date.

The current and new MDP may:

(h) Object to the change request in accordance with the objection rules provided in clause 2.10.

2.7 RP Obligations

The new RP must:

- (a) Change the name of the current RP using change reason code 6300 if this change has not been performed by the new FRMP.
- (b) Ensure that the name of the new MPB is recorded in the MSATS system, using change reason code 6700, prior to the actual change date.
- (c) Ensure that the name of the MDP is recorded in the MSATS system, using change reason code 6200, prior to the actual change date.

The new RP may:

- (d) Provide a change request with its participant identifier, and the names of the MPB and MPC.
- (e) Object to the change request in accordance with the objection rules provided in clause 2.10.

2.8 MPB Obligations

The MPB must:

(a) On receipt of a data request for change reason codes 1000 or 1020, initiate a change request to provide some or all of the following information within 2 business days of receipt of the data request:

Meter Installation Type Code	Meter Serial Number	Meter Status				
Register ID	Register ID Status	Actual Cumulative Indicator				
Dial Format	Multiplier	Network Tariff Code				
Time Of Day	Unit of Measure	Controlled Load				

The current and new MPB may

(b) Object to the change request in accordance with the objection rules in clause 2.10.

2.9 Time Frame Rules

(a) The valid time frame rules for each jurisdiction for the respective change reason codes are shown in the Table 2-A.

(b) When preparing an initial change request, the new FRMP must choose, for any given change reason code, a proposed change date that lies between the prospective business day and the retrospective business day limits.

Table 2-A - Time frame rules

CR Code 1000 – Change Retailer										
Jurisdiction	Objection Logging Business Days	Objection Clearing Business Days	Retrospective Business Days	Prospective Business Days						
ALL	1	0 65								
CR Code 10	020 – Change Ret	ailer-Retro-Long Term	/ Error (not SMALL	-)						
Jurisdiction	, 00 0	Objection Clearing Business		Prospective Business						
	Business Days	Days	Business Days	Days						

2.10 Objection Rules

(a) The objection for the change reason codes identified in clause 2.1 are specified in Table 2-B.

Table 2-B - Objection rules**

	10010 2															
	CR Cod	le 1000 -	Cha	nge R	etail	er										
Objection	NMI	la curlon	FRI	MP	LR		MDF	•	MPE	3	RO	LR	RP		LN	SP
Code	Class	Jur'n	N	С	N	C	N	С	N	С	Ν	C	N	С	N	С
BADMETER	ALL	ALL	-	-	-	-	Yes	Yes	Yes	Yes	-	-	Yes	-	-	Yes
BADPARTY	ALL	ALL	-	-	-	-	-	-	-	-	-	-	Yes	-	-	-
DECLINED	ALL	ALL	-	-	-	-	Yes	-	-	-	-	-	Yes	-	-	-
NOTAPRD	ALL	ALL	-	-	-	-	-	-	-	-	-	-	-	-	-	Yes
DATEBAD	ALL	ALL	-	-	-	-	-	-	-	-	-	-	-	-	-	-
DEBT	ALL	QLD	-	Yes	-	-	-	-	-	-	-	-	-	-	-	-
CONTRACT	ALL	QLD	-	Yes	-	-	-	-	-	-	-	-	-	-	-	-
	CR Cod	le 1020 –	Cha	nge R	etail	er-R	etro-L	ong i	Term /	'Erroi	r (not	SM	ALL)			
Objection	NMI	la contra	FRMP LI		LR	LR MDP		MPB RO		ROLR RP		LNSP				
Code	Class	Jur'n	N	С	N	С	N	С	N	С	N	С	N	С	N	С
BADMETER	ALL	ALL	-	-	-	-	Yes	Yes	-	-	-	-	Yes	-	-	Yes
BADPARTY	ALL	ALL	-	-	-	-	-	-	-	-	-	-	Yes	-	-	-
DECLINED	ALL	ALL	-	-	-	-	Yes	-	-	-	-	-	Yes	-	-	-
NOTAPRD	ALL	ALL	-	-	-	-	-	-	-	-	-	-	-	-	-	Yes
DATEBAD	ALL	ALL	-	-	-	-	Yes	Yes	-	-	-	-	-	-	-	-
RETRO	ALL	ALL	-	Yes	-	-	-	-	-	-	-	-	-	-	-	-
DEBT	ALL	QLD	-	Yes	-	-	-	-	-	-	-	-	-	-	-	-
CONTRACT	ALL	QLD	-	Yes	-	-	-	-	-	-	-	-	-	-	-	-

^{**} N = New role, C = Current role.

2.11 Change Request Status Notification Rules

(a) The valid change request status notification rules for the change reason codes identified in clause 2.1 are specified in Table 2-C.

Table 2-C – Change request status notification rules**

CR Code 1000 – Change Retailer CR Code 1020 – Change Retailer - Retrospective - Long Term / Error PARTICIPANT ROLE & ROLE STATUS – Receives Notification of Change

Status	FRMP		LR		Ľ	SP	MDF	•	MPE	3	Rol	_R	NS	P2	RP	
Change	N	С	Ν	С	N	С	N	С	N	С	N	С	N	С	N	С
CANCELLED	Yes	Yes	-	-	-	Yes	Yes	Yes	Yes	Yes	-	-	-	-	Yes	Yes
COMPLETED	Yes	Yes	-	•	•	Yes	Yes	Yes	Yes	Yes		-	•	Yes	Yes	Yes
OBJECTED	Yes	Yes		-	-	Yes	Yes	Yes	Yes	Yes	-	-	-	-	Yes	Yes
PENDING	Yes	Yes	-	•	•	Yes	Yes	Yes	Yes	Yes		-	•	-	Yes	Yes
REJECTED	Yes	Yes	ı	•	•	Yes	Yes	Yes	-	•	1	-	•	•	Yes	Yes
REQUESTED	Yes	Yes	-		•	Yes	Yes	Yes	Yes	Yes	-	-		-	Yes	Yes

^{**} N = New role, C = Current role.

3. CHANGE FRMP – WHERE FRMP IS NOT THE RP

3.1 Application [1050 1051]

This procedure applies to the following change reason codes:

• 1050 - Change Retailer - FRMP not RP (not SMALL)

This is the situation where a new FRMP obtains the consent of the consumer to take responsibility for the supply of electricity to that connection point. The date of transfer would be a nominal prospective date. The NMI could transfer from a Tier-1 site to a Tier-2 site, or from a Tier-2 site to another Tier-2 site, or from a Tier-2 site to a Tier-1 site.

• 1051 - Change Retailer - Retrospective - FRMP not RP (not SMALL)

This is the situation where a new FRMP obtains the consent of the consumer to take responsibility for the supply of electricity to that connection point. The date of transfer could be a date agreed by the current and new FRMP (a retrospective date) where the actual transfer date was in error. The NMI could transfer from a Tier-1 site to a Tier-2 site, or from a Tier-2 site to another Tier-2 site, or from a Tier-2 site to a Tier-1 site.

3.2 Conditions Precedent

- (a) The NMI already exists in MSATS. If the NMI does not already exist in MSATS, the retailer should refer to chapter 5 "Create a NMI" and seek LNSP action to establish the NMI.
- (b) The NMI classification is wholesale, interconnector, generator or sample.
- (c) Chapter 3 relates only to change reason codes 1050 or 1051.
- (d) The date of the transfer of retailer from the current FRMP to the new FRMP will be the actual change date.
- (e) The actual change date is to be established in accordance with the read type code submitted to the MSATS system by the new FRMP.
- (f) The new FRMP is not the new RP.

3.3 Initiation Rules

- (a) A new FRMP may initiate a change request to effect a change of retailer in accordance with clause 3.4.
- (b) The new FRMP must use one of the following change reason codes 1050 or 1051 to establish an initial change request.

3.4 FRMP Obligations

The new FRMP must:

- (a) Comply with the consumer consent provisions of the relevant jurisdiction for the NMI.
- (b) Obtain the NMI checksum from an approved source.
- (c) Confirm that the NMI is a valid NMI for the connection point prior to the initiation of a change request.
- (d) Submit a change request to CATS to initiate the transfer of the consumer for which the new retailer has obtained an explicit informed consent.
- (e) Populate the initial change request with the following information:

Change reason code	Participant transaction ID	NMI and NMI checksum
CATS participant ID	Proposed change date	RP
MDP	Read type code	

(f) Nominate itself as the new FRMP.

The new FRMP may:

(g) Populate the initial change request with the identity of the following roles:

MPB	MPC	
-----	-----	--

or leave this action to the new RP to complete.

(h) For change reason code 1051 (retrospective), populate the initial change request with:

Actual end date	
Actual ella date	

- (i) Nominate, in accordance with clause 3.4(e), a proposed change date, which is retrospective to the date on which the change request was initiated. The limits for retrospective transfer dates are provided in clause 3.9.
- (j) Withdraw a change of retailer transaction request at any time until the transfer is completed in CATS.

The current FRMP may:

(k) Object to the change request in accordance with the objection rules provided in clause 3.10.

3.5 LNSP Obligations

The current LNSP must:

- (a) Provide the new MDP with the average daily load on request of that MDP.
- (b) In the case of a move-in, verify that the site is energised and if not facilitate the energisation of the site on the proposed transfer date or up to 2 days early. It is expected that requests for energisation will be made through a separate process between the retailer and the LNSP to ensure that the transfer process does not delay the energisation of the site for the consumer.
- (c) Respond to a request from the MSATS system for data within 1 day of the date of the data request.

The current LNSP may:

(d) Object to the change request in accordance with the objection rules provided in clause 3.10.

3.6 MDP Obligations

The new MDP must:

- (a) Once the transfer has completed, set up the NMI suffix(s) so that they become active on the actual change date.
- (b) On receipt of a data request for change reason codes 1050 or 1051 initiate a change request to provide the actual change date, which must be the date nominated as the proposed date by the new FRMP unless the date nominated is inappropriate, in which case the MDP must advise the new FRMP of the reason why the proposed date is inappropriate and mutually agree on what the actual change date should be.
- (c) Validate that the metering installation type code is identified as a type 1, type 2, type 3 or type 4 metering installation. If an error is identified, advise the new FRMP and await instructions.
- (d) For prospective changes that do not require a meter read, provide the actual change date within 2 days of the actual change date being reached.
- (e) For changes that require a meter read, provide the actual change date within 2 days of the meter read.
- (f) For retrospective changes where the rules specify that an Actual Change Date is required, ensure that the actual change date submitted is no earlier than the time limit specified by the jurisdiction for the change reason code on the change request.
- (g) For example, if the jurisdictional time frame for a change reason code is 10 days, then the actual change date must be no earlier than 10 days prior to the date the data request was received, even if MSATS would allow an earlier date.

The new MDP may:

(h) Object to the change request in accordance with the objection rules provided in clause 3 10

3.7 RP Obligations

The new RP must:

(a) Ensure that the name of the new MPB is recorded in the MSATS system, using change reason code 6700, prior to the actual change date.

The new RP may:

- (b) Provide a change request with its participant identifier, and the names of the MPB and MPC.
- (c) Object to the change request in accordance with the objection rules provided in clause 3.10.

3.8 MPB Obligations

The MPB must:

(a) On receipt of a data request for change reason codes 1050 or 1051, initiate a change request to provide some or all of the following information within 2 business days of receipt of the data request:

Meter Installation Type Code	Meter Serial Number	Meter Status
Register ID	Register ID Status	Actual Cumulative Indicator
Dial Format	Multiplier	Network Tariff Code
Time Of Day	Unit of Measure	Controlled Load

3.9 Time Frame Rules

- (a) The valid time frame rules for each jurisdiction for the respective change reason codes are shown in the Table 3-A.
- (b) When preparing an initial change request, the new FRMP must choose, for any given change reason code, a proposed change date that lies between the prospective business day and the retrospective business day limits.

Table 3-A - Time frame rules

CR Code 1050 – Change Retailer - FRMP not RP (not SMALL)									
Jurisdiction	Objection Logging	Objection Clearing Business	Retrospective	Prospective Business					
	Business Days	Days	Business Days	Days					
ALL	1	20	0	65					
CR Code 1051 – Change Retailer - Retrospective - FRMP not RP (not SMALL)									
CR Code 10)51 – Change Ret	ailer - Retrospective - F	RMP not RP (not	SMALL)					
CR Code 10 Jurisdiction		ailer - Retrospective - F Objection Clearing Business	•	SMALL) Prospective Business					
			•	·					

3.10 Objection Rules

(a) The valid objection rules for the change reason codes identified in clause 3.1 are specified in Table 3-B.

Table 3-B - Objection rules for change reason codes**

	Table 3-b -	able 3-B – Objection rules for change reason codes														
	CR Code 1	050 – Cha	ange	Reta	iler ·	- FR	MP no	ot RP	(not S	SMAL	L)					
Objection	NMI	la contra	FR	MP	LR		MDF	•	MPE	3	RO	LR	RP		LN	SP
Code	Class	Jur'n	N	С	Ν	С	N	С	N	С	Ν	C	N	С	Ν	С
BADMETER	ALL	ALL	-	-	-	-	Yes	Yes	Yes	Yes	-	-	Yes	-	-	Yes
BADPARTY	ALL	ALL	-	-	-	-	-	1	1	-	-	-	Yes	-	-	1
DECLINED	ALL	ALL	-	-	-	-	Yes	-	-	-	-	-	Yes	-	-	-
NOTAPRD	ALL	ALL	-	-	-	-	-	-	-	-	-	-	-	-	-	Yes
DATEBAD	ALL	ALL	-	-	-	-	Yes	-	-	-	-	-	-	-	-	-
DEBT	ALL	QLD	-	Yes	-	-	-	-	-	-	-	-	-	-	-	-
CONTRACT	ALL	QLD	-	Yes	-	-	-	-	-	-	-	-	-	-	-	-
	CR Code 1	051 – Cha	ange	Reta	iler ·	- Re	trospe	ective	- FRI	MP no	t RP	(no	t SMA	LL)		
Objection	NMI	1	FR	MP	LR		MDF	•	MPE	3	RC	LR	RP		LN	SP
Code	Class	Jur'n														
	Class		N	С	Ν	С	N	С	N	С	N	С	N	С	N	С
BADMETER	ALL	ALL	N -	C -	N -	C			N -	C -	N -	C	N Yes	C	N -	C Yes
BADMETER BADPARTY		ALL ALL	- -	- -	- -	C	N	С	N -	C	N - -	- -		_		_
	ALL		- -	- -	- -	- -	N	С	- -	- - -	N - -	- -	Yes	_		_
BADPARTY	ALL ALL	ALL	-	- - -	- - -	- - -	N Yes	C Yes	- - -	- - -	-	- - -	Yes Yes	-	-	Yes -
BADPARTY DECLINED	ALL ALL ALL	ALL	-	- - - -	N	- - - -	N Yes	C Yes	- - - -	- - - -	-	- - - -	Yes Yes	-	-	Yes - -
BADPARTY DECLINED NOTAPRD	ALL ALL ALL	ALL ALL ALL	- - -	- - - - - Yes	- - - -	- - - -	N Yes - Yes	C Yes	- - - -	- - - -	-	- - - -	Yes Yes	-	-	Yes - -
BADPARTY DECLINED NOTAPRD DATEBAD	ALL ALL ALL ALL ALL	ALL ALL ALL ALL	- - - -	- - - -	N	- - - -	N Yes - Yes	C Yes	N	- - - - -	-	- - - - -	Yes Yes	-	-	Yes - -

N = New role, C = Current role.

3.11 Change Request Status Notification Rules

(a) The valid change request status notification rules for the change reason codes identified in clause 3.1 are specified in Table 3-C.

Table 3-C - Change request status notification rules**

CR Code 1050 - Change Retailer - FRMP not RP (not SMALL) CR Code 1051 - Change Retailer - Retrospective - FRMP not RP (not SMALL) **Status Change FRMP** LR **LNSP MDP MPB RoLR** NSP₂ N C C C C C C С Ν Ν Ν Ν Ν C Ν N CANCELLED Yes Yes Yes Yes Yes Yes Yes Yes Yes **COMPLETED** Yes Yes Yes Yes Yes Yes Yes Yes -Yes Yes OBJECTED Yes Yes Yes Yes Yes Yes Yes -Yes Yes PENDING Yes Yes Yes Yes Yes Yes Yes Yes Yes **REJECTED** Yes Yes Yes Yes Yes Yes Yes Yes **REQUESTED** Yes Yes Yes Yes Yes Yes Yes Yes

^{**} N = New role, C = Current role.

4. PROVIDE DATA – PROVIDE ACTUAL CHANGE DATE

4.1 Application [1500]

This procedure applies to the following change reason code:

• 1500 - Provide Actual Change Date

This is the situation where the MDP is required to provide the MSATS system with the actual change r date that specifies the date on which the change on the related change request will take effect. Conditions Precedent

- (a) The NMI exists in the MSATS system.
- (b) The FRMP has initiated a change request.
- (c) The NMI classification is wholesale, interconnector, generator or sample.
- (d) The MDP has received a data request from the MSATS system for the provision of the actual change date.
- (e) There are no objections allowed for this change reason code and NMI classification.
- (f) Chapter 4 relates only to change reason code 1500.

4.2 Initiation Rules

- (a) The MDP may initiate a change request to submit an actual change date to the MSATS system in accordance with clause 4.3.
- (b) The MDP must use change reason code 1500 to provide the actual change date.

4.3 MDP Obligations

The MDP must:

- (a) Obtain the NMI checksum from an approved source.
- (b) Confirm that the NMI is a valid NMI for the connection point prior to the initiation of a change request.
- (c) Provide the Actual Change Date to MSATS within 2 days of reading the meter or otherwise prescribed by the relevant jurisdiction. In the case of a retrospective transaction, it is to be provided within 2 days of notification.
- (d) Populate an initial change request with the following information:

Change reason code	Participant transaction ID	NMI and NMI checksum
CATS participant ID	Actual change date	Related change request ID
MDP		

4.4 Time Frame Rules

(a) The valid time frame rules for each jurisdiction for the change reason code are shown in the Table 4-A. Note that the time frame rules in Table 4-A do not override the MDP obligation in clause 4.3(c).

Table 4-A - Time frame rules

CR Code 15	CR Code 1500 – Provide actual change date										
Jurisdiction	Objection Logging	Objection Clearing Business	Retrospective	Prospective Business							
	Business Days	Days	Business Days	Days							
ALL	0	0	20	0							

4.5 Objection Rules

(a) There are no objections.

4.6 Change Request Status Notification Rules

(a) The valid change request status notification rules for the change reason codes identified in clause 4.1 are specified in Table 4-B.

Table 4-B – Change request status notification rules**

		CR Code 1500 – Provide Actual Change Date PARTICIPANT ROLE & ROLE STATUS – Receives Notification of Change														
Status	FRMF	,	LR		LNS	SP	MDP		MP	В	Rol	.R	NSF	2	RP	
Change	N	С	N	С	N	С	N	С	N	С	N	С	N	С	N	С
CANCELLED	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
COMPLETED	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
OBJECTED	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
PENDING	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
REJECTED	-	-	-	-	-	-	Yes	Yes	-	-	-	-	-	-	-	-
REQUESTED	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-

^{**} N = New role, C = Current role.

5. CREATE NMI - CREATE A NMI

5.1 Application [2000 2001]

This procedure applies to the following change reason codes:

2000 – Create NMI Details

This is the situation where the new LNSP establishes the initial set of information in the MSATS system in regard to a connection point. The date at which the information will apply would be a prospective date.

• 2001 - Create NMI Details - Retrospective

This is the situation where the new LNSP establishes the initial set of information in the MSATS system in regard to a connection point. The date at which the information will apply would be a retrospective date.

5.2 Conditions Precedent

- (a) The NMI does not exist in MSATS.
- (b) The NMI classification is wholesale, interconnector, generator or sample.
- (c) Chapter 5 relates only to change reason codes 2000 and 2001.

5.3 Initiation Rules

- (a) A new LNSP may initiate a change request to create a NMI record in the MSATS system in accordance with clause 5.4.
- (b) The new LNSP must use one of the following change reason codes 2000 or 2001 to establish an initial change request.

5.4 LNSP Obligations

The new LNSP must:

- (a) Obtain the NMI checksum from an approved source.
- (b) Confirm that the NMI is a valid NMI for the connection point prior to the initiation of a change request.
- (c) Populate an initial change request with the following information:

Participant transaction ID	NMI and NMI checksum	CATS participant ID
Proposed change date	TNI code	DLF code
NMI classification code	Jurisdiction code	FRMP
LR	ROLR	RP
MDP	МРВ	MPC
NMI status code	Locality	State
Postcode	Change reason code	

- (d) Nominate itself as the new LNSP.
- (e) Populate the initial change request with values for the following address fields (as applicable):

EITHER

DPID	Flat number	Flat type
Floor number	Floor type	House number
House number suffix	Location descriptor	Lot number
Street name	Street suffix	Street type

OR

d address 1 Unstructured address 2 Unstructured address 3

The new LNSP may:

(f) Populate the initial change request with the following information:

Embedded network	Building name	
parent name		

5.5 MPB Obligations

The new MPB must:

(a) On receipt of the change request status notification for the completed status, provide the metering installation type code to the MSATS system using change reason code 3000 or 3001.

5.6 Time Frame Rules

(a) The valid time frame rules for each jurisdiction for the respective change reason codes are shown in the Table 5-A.

Table 5-A - Time frame rules

CR Code	2000 – Create N	//I Details			
Jurisdiction					Prospective Business
	Business Days	Days	Bl	usiness Days	Days
ALL	1	20		0	65
CR Code	2001 - Create N	MI Details- Retrospective			
Jurisdiction	Objection Logging	Objection Clearing Business D	ays (Retrospective	Prospective
	Business Days	-	-	Business Days	Business Days
ALL	1	20		130	0

5.7 Objection Rules

(a) The valid objection rules for the change reason codes identified in clause 5.1 are specified in Table 5-B.

Table 5-B - Objection rules**

	CR Code	2000 –	Create	NMI	Details											
Objection	NMI	luwin.	FRM	•	LR		MDP		MPB		ROLF	₹	RP		LNS	SP
Code	Class	Jur'n	N	С	N	С	N	С	N	C	N	С	N	С	Ν	С
NOTRESP	ALL	ALL	Yes	-	Yes	-	Yes	-	Yes	-	Yes	-		-	-	-
BADPARTY	ALL	ALL	-	-	-	-	-	-	-	-	-	-	Yes	•	-	-
	CR Code	2001 –	Create	NMI	Details	- R	etro									
Objection	NMI	Leader	FRM	•	LR		MDP		MPB		ROLF	₹	RP		LNS	SP
Code	Class	Jur'n	N	C	N	C	N	C	N	С	N	С	N	С	N	С
NOTRESP	ALL	ALL	Yes	-	Yes	-	Yes	-	Yes	-	Yes	-	-	-	-	-
BADPARTY	ALL	ALL	-	-	-	-	-	-	-	-	-	-	Yes	-	-	-
RETRO	ALL	ALL	Yes	-	Yes	-	-	-		-	-	-	-	-	-	-

^{**} N = New role, C = Current role.

5.8 Change Request Status Notification Rules

(a) The valid change request status notification rules for the change reason codes identified in clause 5.1 are specified in Table 5-C.

Table 5-C – Change request status notification rules**

	CR C	ode	2001 –	Crea	ate NM ate NM & ROI	l Det	ails - R		eceives	s Not	ificat	tion (of Cha	nge
Status Change	FRMI	P	LR		LNSF	•	MDP		MPB		Rol	_R	RP	
	N	С	N	С	N	С	N	С	N	C	N	C	N	С
CANCELLED	Yes	-	Yes	-	Yes	-	Yes	-	Yes	1	-	1	Yes	-
COMPLETED	Yes	-	Yes	-	Yes	-	Yes	-	Yes		-		Yes	-
OBJECTED	Yes	-	Yes	-	Yes	-	Yes	-	Yes	-	-	-	Yes	-
PENDING	Yes	-	Yes	-	Yes	-	Yes	-	Yes	-	-	-	Yes	-
REJECTED	Yes	-	Yes	-	Yes	-	Yes	-	Yes	-	-	-	Yes	-
REQUESTED	Yes	-	Yes	-	Yes	-	Yes	-	Yes	-	-	-	Yes	-

^{**} N = New role, C = Current role.

6. CREATE NMI – CREATE NMI, METERING INSTALLATION DETAILS AND MDM DATASTREAM

6.1 Application [2500 2501]

This procedure applies to the following change reason codes:

• 2500 - Create NMI, Datastream + Metering Details

This is the situation where the new LNSP is able to provide the datastream and metering installation details at the same time it establishes the initial set of information in the MSATS system in regard to a connection point. The date at which the information will apply would be a prospective date.

2501 – Create NMI, Datastream + Metering Details - Retro

This is the situation where the new LNSP is able to provide the datastream and metering installation details at the same time it establishes the initial set of information in the MSATS system in regard to a connection point. The date at which the information will apply would be a retrospective date.

6.2 Conditions Precedent

- (a) The NMI does not exist in MSATS.
- (b) The NMI classification is wholesale, interconnector, generator or sample.
- (c) Chapter 6 relates only to change reason codes 2500 and 2501.

6.3 Initiation Rules

- (a) If the new LNSP meets the condition imposed by clause 6.2, then the new LNSP may initiate a change request to create a NMI, metering installation details and MDM datastream in the MSATS system in accordance with clause 6.4.
- (b) The new LNSP must use one of the following change reason codes 2500 or 2501 to establish an initial change request.

6.4 LNSP Obligations

The new LNSP must:

- (a) Obtain the NMI checksum from an approved source.
- (b) Confirm that the NMI is a valid NMI for the connection point prior to the initiation of a change request.
- (c) Populate an initial change request with the following information:

Change reason code	Participant transaction ID	NMI and NMI checksum
CATS participant ID	Proposed change date	TNI code
DLF code	NMI classification code	Jurisdiction code
FRMP	LR	ROLR
RP	MDP	MPB
MPC	NMI status code	Locality
State	Postcode	Meter serial ID (at least one)
Metering installation type (for each meter)	Meter status (for each meter)	Register ID (at least one for each meter)
Register ID status (for each register)	Network tariff code (for each register)	Actual / cumulative indicator (for each register)
Controlled load indicator (for each register)	Dial format (for each register)	Multiplier value (for each register)
Time of day code (for each register)	Unit of measure code (for each register)	NMI suffix (at least one)
Datastream type (for each suffix)	Profile name (for each suffix)	Daily average load (for each suffix)
Datastream status code (for each suffix)		

- (d) Nominate itself as the new LNSP.
- (e) Provide the average daily load if this value is not otherwise provided by the new FRMP.
- (f) Populate the initial change request with values for the following address fields (as applicable):

EITHER

DPID	Flat number	Flat type
Floor number	Floor type	House number
House number suffix	Location descriptor	Lot number
Street name	Street suffix	Street type

OR

Unstructured address 1	Unstructured address 2	Unstructured address 3

The new LNSP may:

(g) Populate the initial change request with the following information:

Embedded network	Building name	
parent name		

(h) Populate the initial change request with the following information for each meter:

Additional site information	Next scheduled read date	Meter location
Meter hazard	Meter route	Meter use
Meter point	Meter manufacturer	Meter model
Transformer location	Transformer type	Transformer ratio
Meter constant	Last test date	Next test date
Test result accuracy	Test result notes	Test performed by
Measurement type	Meter program	Meter read type
Remote phone number	Communication equipment type	Communication protocol
Data conversion arrangements	Data validation arrangements	Estimation instructions
Asset management plan details	Calibration tables (details of any calibration factors programmed into the meter)	Password details (the read and time set passwords only, separated by a space; the write password is not to be recorded in MSATS)
Test and calibration program details	User access rights details (i.e. details of any enduse customer access to the metering installation such as pulse outputs)	

(i) Populate the initial change request with the following information for each register:

Network tariff code additional information	Demand value 1, if the network tariff charge includes a demand component	Demand value 2, if the network tariff charge includes a second demand component
MDM contributory suffix (this can also be supplied by the MDP)		

6.5 Time Frame Rules

(a) The valid time frame rules for each jurisdiction for the respective change reason codes are shown in the Table 6-A.

Table 6-A - Time frame rules

CR Code 25	500 - Create NMI,	Datastream + Metering	Details	
Jurisdiction	Objection Logging	Objection Clearing Business	Retrospective	Prospective Business
	Business Days	Days	Business Days	Days
ALL	1	20	0	65
CR Code 25	501 - Create NMI,	Datastream + Metering	Details – Retrosp	ective
CR Code 25 Jurisdiction		Datastream + Metering Objection Clearing Business	•	ective Prospective Business
			•	

6.6 Objection Rules

(a) The valid objection rules for the change reason codes identified in clause 6.1 are specified in Table 6-B.

Table 6-B - Objection rules**

	CR Code 25	500 – Cr	eate I	νMΙ,	Datas	strea	am + M	Vlete	r Deta	ails						
Objection	NIMI Class	Lunda	FRN	1P	LR		MDF	•	MPE	3	ROL	.R	RP		LNS	SP
Code	NMI Class	Jur'n	N	С	N	ပ	N	С	N	С	N	С	N	С	N	С
NOTRESP	ALL	ALL	Yes	-	Yes	-	Yes	-	Yes	-	Yes	-	Yes	-	-	-
BADMETER	ALL	ALL	-	-	-	-	Yes	-	-	-	-	-	-	-	-	-
CR Code 2501 - Create NMI, Datastream + Meter Details - Retro																
Objection	NIMI OL		FRM	IP	LR		MDF	•	MPE	3	ROL	.R	RP		LNS	SP
Code	NMI Class	Jur'n	N	С	N	С	N	С	N	С	N	С	N	С	N	С
NOTRESP	ALL	ALL	Yes	-	Yes	-	Yes	-	Yes	-	Yes	-	Yes	-	-	-
BADMETER	ALL	ALL	-	-	-	-	Yes	-	-	-	-	-	-	-	-	-
BADPARTY	ALL	ALL	-	-	-	-	-	-	-	-	-	-	Yes	-	-	-
RETRO	ALL	ALL	Yes	-	Yes	-	-	-	-	-	-	-	-	-	-	-

N = New role, C = Current role.

6.7 Change Request Status Notification Rules

(a) The valid change request status notification rules for the change reason codes identified in clause 6.1 are specified in Table 6-C.

Table 6-C - Change request status notification rules**

	CR C	CR Code 2500 – Create NMI, Datastream + Meter Details CR Code 2501 – Create NMI, Datastream + Meter Details - Retro PARTICIPANT ROLE & ROLE STATUS – Receives Notification of Change														
Status Change	FRM	FRMP LR LNSP MDP MPB RoLR NSP2								RP						
	N	С	N	С	N	С	N	С	N	С	N	С	Ν	С	N	С
CANCELLED	Yes	-	Yes	-	Yes	1	Yes	1	Yes	1	-	-	1	-	Yes	-
COMPLETED	Yes	-	Yes	-	Yes	1	Yes	•	Yes	ı		-	ı	-	Yes	-
OBJECTED	Yes	-	Yes	-	Yes	ı	Yes	1	Yes	ı	•	-	ı	1	Yes	-
PENDING	Yes	-	Yes	-	Yes	•	Yes	•	Yes	•	•	-	•	-	Yes	-
REJECTED	Yes	-	Yes	-	Yes	ı	Yes	1	Yes	ı	•	-	ı	1	Yes	-
REQUESTED	Yes	-	Yes	-	Yes	-	Yes	-	Yes	-	-	-	-	-	Yes	-

^{**} N = New role, C = Current role.

7. MAINTAIN PROFILE SHAPES – CREATE AND CHANGE EXTERNAL PROFILE SHAPE

7.1 Application [2100 2101 5110 5111]

This procedure applies to the following change reason codes:

• 2100 - Create External Profile Shape

This is the situation where AEMO in accordance with the *Metrology Procedure* (Part A & B) is required to provide the MSATS system with a name of an external profile shape. The date at which the information will apply would be a prospective date.

• 5110 - Change External Profile Shape

This is the situation where AEMO in accordance with the *Metrology Procedure* (Part A & B) is required to change details associated with an external profile shape which had been recorded in the MSATS system. The date at which the information will apply would be a prospective date.

2101 – Create External Profile Shape - Retro

This is the situation where AEMO in accordance with the *Metrology Procedure* (Part A & B) is required to provide the MSATS system with a name of an external profile shape. The date at which the information will apply would be a retrospective date.

5111 – Change External Profile Shape - Retro

This is the situation where AEMO in accordance with the *Metrology Procedure* (Part A & B) *is* required to change details associated with an external profile shape which had been recorded in the MSATS system. The date at which the information will apply would be a retrospective date.

7.2 Conditions Precedent

- (a) For codes 5110 and 5111, the NMI already exists in MSATS. For codes 2100 and 2101, the NMI does not exist in MSATS.
- (b) The NMI classification is EPROFILE.
- (c) There are no objections allowed for these change reason codes.
- (d) Chapter 7 relates only to change reason codes 2100, 2101, 5110, and 5111.

7.3 Initiation Rules

- (a) The new 'NEMM' role may initiate a change request to create an external profile shape in accordance with clause 7.4.
- (b) The new 'NEMM' role must use one of the following change reason codes 2100 or 2101 to establish an initial change request.
- (c) The current 'NEMM' role may initiate a change request to change an external profile shape in accordance with clause 7.4.
- (d) The current 'NEMM' role must use one of the following change reason codes 5110 or 5111 to establish an initial change request.

7.4 AEMO Obligations

AEMO must:

(a) Confirm that the NMI is a valid NMI for the profile shape prior to the initiation of a change request. (b) Populate an initial change request with the following information:

Change reason code	Participant transaction ID	NMI and NMI checksum
CATS participant ID	Proposed change date	

(c) For change reason code 2100 or 2101, populate an initial change request with the following information:

NMI status code	NMI classification code	Jurisdiction code				
DLF code	TNI code	MDP				
FRMP	LR	RP				
ROLR	MPB	LNSP				
MPC	NMI suffix	Datastream type				
Profile name	Datastream status code	Average daily load				
Meter serial number	Meter installation code	Meter status code				

(d) For change reason code 2100 or 2101, nominate itself as the new NEMM.

AEMO may:

(e) For change reason code 5110 or 5111, populate the initial change request with the following information:

NMI status code	Jurisdiction code	FRMP				
LR	LNSP	MPB				
ROLR	RP	DLF code				
TNI code	Aggregate flag	MDP				
NMI suffix	Datastream type	Profile name				
Datastream status code	Average daily load	Meter serial number				
Meter installation code	Meter status code					

(f) For change reason code 5110 or 5111, nominate itself in the role of NEMM.

7.5 Time Frame Rules

(a) The valid time frame rules for each jurisdiction for the respective change reason codes are shown in the Table 7-A.

Table 7-A - Time frame rules

CR Code 2100 – Create External Profile Shape CR Code 5110 – Change External Profile Shape									
Jurisdiction	Objection Logging Business Days	Objection Clearing Business Days	Retrospective Business Days	Prospective Business Days					
ALL 1 10 0 65									
CR Code 2101 – Create External Profile Shape - Retro CR Code 5111 – Change External Profile Shape - Retro									
	l11 – Change Ext			Prospective Business Days					

7.6 Objection Rules

(a) The valid objection rules that for the change reason codes identified in clause 7.1 are specified in Table 7-B.

Table 7-B - Objection rules for change reason codes**

CR Code 2100 – Create External Profile Shape CR Code 2101 – Create External Profile Shape - Retro																
Objection	NMI	IMI		MP	LR		MDF	•	MPE	3	RO	LR	RP		LN	SP
Code	Class	Jur'n	N	С	N	С	N	С	N	С	N	С	N	С	N	С
NOTRESP	ALL	ALL	-	-	-	-	Yes	-	-	-	-	-	-	-	-	-
	CR Code 5110 – Change External Profile Shape CR Code 5111 – Change External Profile Shape - Retro															
Objection	NMI	la color	FRI	MP	LR		MDF	•	MPE	3	RO	LR	RP		LN	SP
Code	Class	Jur'n	N	С	N	С	N	С	N	С	N	С	N	С	N	С
DECLINED	ALL	ALL			_		Yes									

N = New role, C = Current role.

7.7 Change Request Status Notification Rules

(a) The valid change request status notification rules for the change reason codes identified in clause 7.1 are specified in Table 7-C.

Table 7-C - Change request status notification rules**

	CR C	ode 21	l 01 –	Crea	ate Ex		Profi	le Sh	ape - Re	etro Notifica	tion o	f Cha	ınge			
Status	NEM	NEMM FRMP		LR	LR LNSP		MDP		MP	В	ROLR		RP			
Change	N	С	Ν	Υ	N	С	N	С	N	С	N	С	N	С	N	С
CANCELLED	Yes	-	-	-	-	-	-	-	Yes	-	-	-	-	-	-	-
COMPLETED	Yes	-	-	-	-	-	-	-	Yes	-	-	-	-	-	-	-
OBJECTED	Yes	-	-	-	-	-	-	-	Yes	-	-	-	-	-	-	-
PENDING	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
REJECTED	Yes	-	-	-	-	-	-	-	Yes	-	-	-	-	-	-	-
REQUESTED	Yes	-	-	-	-	-	-	-	Yes	-	-	-	-	-	-	-
	CR Code 5110 – Change External Profile Shape															

CR Code 5111 – Change External Profile Shape - Retro
PARTICIPANT ROLE & ROLE STATUS – Receives Notification of Change

Status	NEM	M	FR	MP	LR		LNS	SP	MDP		MPI	В	ROI	LR	RP	
Change	N	С	N	С	N	С	N	С	N	С	N	С	N	С	Ν	С
CANCELLED	-	Yes	-	-	-	-	-	-	Yes	Yes	-	-	-	-	-	-
COMPLETED	-	Yes	-	-	-	-	-	-	Yes	Yes	-	-	-	-	-	-
OBJECTED	-	Yes	-	-	-	-	-	-	Yes	Yes	-	-	-	-	-	-
PENDING	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
REJECTED	-	Yes	-	-	-	-	-	-	Yes	Yes	-	-	-	-	-	-
REQUESTED	-	Yes	-	-	-	-	-	-	Yes	Yes	-	-	-	-	-	-

^{**} N = New role, C = Current role.

8. MAINTAIN METERING – CREATE METERING INSTALLATION DETAILS

8.1 Application [3000 3001]

This procedure applies to the following change reason codes:

• 3000 - Create Metering Installation Details

This is the situation where the current MPB establishes the initial set of information in the MSATS system in regard to the metering installation details, and there is no associated meter removal for the NMI for the same effective date. The date at which the information will apply would be a prospective date.

• 3001 - Create Metering Installation Details - Retrospective

This is the situation where the current MPB establishes the initial set of information in the MSATS system in regard to the metering installation details, and there is no associated meter removal for the NMI for the same effective date. The date at which the information will apply would be a retrospective date.

8.2 Conditions Precedent

- (a) The NMI exists in the MSATS system.
- (b) The metering installation details do not exist in MSATS.
- (c) The NMI classification is wholesale, interconnector, generator or sample.
- (d) There are no objections allowed for this change reason code and NMI classification.
- (e) Chapter 8 relates only to change reason codes 3000 and 3001.

8.3 Initiation Rules

- (a) A current MPB may initiate a change request to create metering installation records in the MSATS system in accordance with clause 8.4.
- (b) The current MPB must use one of the following change reason codes 3000 or 3001 to establish an initial change request.

8.4 MPB Obligations

The MPB must:

- (a) Obtain the NMI checksum from an approved source.
- (b) Confirm that the NMI is a valid NMI for the connection point prior to the initiation of a change request.
- (c) Populate an initial change request with the following information:

Change reason code	Participant transaction ID	NMI and NMI checksum
CATS participant ID	Proposed change date	Meter serial ID (at least one)
Metering installation type (for each meter)	Meter status (for each meter)	Register ID (at least one for each meter)
Register ID status (for each register ID)	Actual / cumulative indicator (for each register)	Controlled load indicator (for each register ID)
Dial format (for each register ID)	Multiplier value (for each register ID)	Time of day code (for each register ID)
Unit of measure code (for each register ID)		

The current MPB may:

(d) Populate the initial change request with the following information for each meter:

Additional site information	Network tariff code (for each register ID)	Next scheduled read date
Meter location	Meter hazard	Meter route
Meter use	Meter point	Meter manufacturer
Meter model	Transformer location	Transformer type
Transformer ratio	Meter constant	Last test date
Next test date	Test result accuracy	Test result notes
Test performed by	Measurement type	Meter program
Meter read type	Remote phone number	Communication equipment type
Communication protocol	Data conversion arrangements	Data validation arrangements
Estimation instructions	Asset management plan details	Calibration tables (details of any calibration factors programmed into the meter)
Password details (the read and time set passwords only, separated by a space; the write password is not to be recorded in MSATS)	Test and calibration program details	User access rights details (i.e. details of any end- use customer access to the metering installation such as pulse outputs)

(e) Populate the initial change request with the following information for each register:

Network tariff code additional information	MDM contributory suffix (this can also be supplied by the MDP)	Demand value 1, if the network tariff charge includes a demand component
Demand value 2, if the network tariff charge includes a second demand component		

(f) For change reason codes that are retrospective, populate the initial change request with:

Actual end date		
-----------------	--	--

8.5 Time Frame Rules

(a) The valid time frame rules for each jurisdiction for the respective change reason codes are shown in the Table 8-A.

Table 8-A - Time frame rules

CR Code 3000 - Create Meter Details									
Jurisdiction	Objection Logging Business Days	Objection Clearing Business Days	Retrospective Business Days	Prospective Business Days					
ALL	0	0	0	65					
CR Code 3001 - Create Meter Details - Retrospective									
Jurisdiction	Objection Logging	Objection Clearing Business	Retrospective	Prospective Business					
	Business Days	Days	Business Days	Days					
ALL	0	0	140	0					

8.6 Objection Rules

(a) There are no objections.

8.7 Change Request Status Notification Rules

(a) The valid change request status notification rules for the change reason codes identified in clause 8.1 are specified in Table 8-B.

Table 8-B - Change request status notification rules**

CR Code 3000 – Create Meter Details
CR Code 3001 – Create Meter Details - Retrospective
PARTICIPANT ROLE & ROLE STATUS – Receives Notification of Change

Status	FRMF	•	LR		LN	SP	MC)P	MP	В	RoL	.R	NSF	2	RP	
Change	N	С	Ν	С	N	С	N	С	N	С	N	C	N	С	N	С
CANCELLED	-	-	-	-	-	-	-	-	-	Yes	-	-	-	-	-	-
COMPLETED	-	Yes	-	Yes	-	Yes	-	Yes	-	Yes	-	-	-	Yes	-	Yes
OBJECTED	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
PENDING	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
REJECTED	-	-	-	-	-	-	-	-	-	Yes	-	-	-	-	-	-
REQUESTED	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-

^{**} N = New role, C = Current role.

9. MAINTAIN METERING – EXCHANGE OF METERING INFORMATION

9.1 Application [3004 3005]

This procedure applies to the following change reason codes:

• 3004 – Exchange of Metering information

This is the situation where the current MPB is required to provide a change to the information in the MSATS system in regard to the metering installation details. The change will include the removal of at least one existing meter and the installation of at least one new meter. A minimum set of metering installation details for the NMI shall exist upon completion of the Change Request. The date at which the information will apply would be a prospective date.

3005 – Exchange of Metering information – Retrospective

This is the situation where the current MPB is required to provide a change to the information in the MSATS system in regard to the metering installation details. The change will include the removal of at least one existing meter and the installation of at least one new meter. A minimum set of metering installation details for the NMI shall exist upon completion of the Change Request. The date at which the information will apply would be a retrospective date.

9.2 Conditions Precedent

- (a) The NMI exists in the MSATS system.
- (b) The metering installation details exist in MSATS.
- (c) At least one meter is removed and one meter is created in each change request.
- (d) The NMI classification is wholesale, interconnector, generator or sample.
- (e) There are no objections allowed for this change reason code and NMI classification.
- (f) The meter serial ID for the installed meter shall be different to the existing meter serial ID for the same NMI
- (g) Chapter 9 relates only to change reason codes 3004 and 3005.

9.3 Initiation Rules

- (a) A current MPB may initiate a change request to change and create metering installation records in the MSATS system in accordance with clause 9.4.
- (b) The current MPB must use one of the following change reason codes 3004 or 3005 to establish an initial change request.
- (c) A minimum set of metering installation details for the NMI shall exist upon completion of the change request.

9.4 MPB obligations

The current MPB must:

- (a) Obtain the NMI checksum from an approved source.
- (b) Confirm that the NMI is a valid NMI for the connection point prior to the initiation of a change request.
- (c) Populate the change request with the following information:

Change reason code	Participant transaction ID	CATS participant ID
Proposed change date	NMI	NMI checksum

Meter serial ID (for each	
meter)	

(d) For all meters associated to the NMI, where the Meter Status Code is to be "C" populate the change request with the following information (where this information does not currently exist in MSATS):

Meter status (for each	Metering installation type	
meter)	(for each meter)	

(e) For all meters associated to the NMI, where the Register ID Status is to be "C" populate the change request with the following information (where this information does not currently exist in MSATS):

Register ID	Unit of measure code (for each register ID)	Multiplier value (for each register ID)
Time of day code (for each register ID)	Dial format (for each register ID)	Register ID status (for each register ID)
Controlled load indicator (for each register ID)	Actual / cumulative indicator (for each register ID)	

The current MPB may:

(f) Populate the change request with the following information for each meter:

Additional site information	Next scheduled read date	Meter location
Meter hazard	Meter route	Meter use
Meter point	Meter manufacturer	Meter model
Transformer location	Transformer type	Transformer ratio
Meter constant	Last test date	Next test date
Test result accuracy	Test result notes	Test performed by
Measurement type	Meter program	Meter read type
Remote phone number	Communication equipment type	Communication protocol
Data conversion arrangements	Data validation arrangements	Estimation instructions
Asset management plan details	Calibration tables (details of any calibration factors programmed into the meter)	Password details (the read and time set passwords only, separated by a space; the write password is not to be recorded in MSATS)
Test and calibration program details	User access rights details (i.e. details of any enduse customer access to the metering installation such as pulse outputs)	MDM contributory suffix (this can also be supplied by the MDP)
Network tariff code additional information	Demand value 1, if the network tariff charge includes a demand component	Demand value 2, if the network tariff charge includes a second demand component
Network Tariff Code (for each register ID)		

(g) For change reason codes that are retrospective, populate the initial change request with:

Actual end date		
-----------------	--	--

- (h) Where changes to Network Tariff information are required:
 - Check that the network tariff code in the MSATS system is correct and, if it is not, update it to have the correct value.
 - Change the network tariff code in the MSATS system to ensure that the current information provided in the MSATS system is the latest information available from the current LNSP.

9.5 Time Frame Rules

(a) The valid time frame rules for each jurisdiction for the respective change reason codes are shown in the Table 9-A.

Table 9-A - Time frame rules

Cr Code 3004 – Exchange of Metering Information							
Jurisdiction	Objection Logging Business Days	Objection Clearing Business Days	Retrospective Business Days	Prospective Business Days			
ALL	0	0	0	65			
Cr Code 3005 – Exchange of Metering Information – Retrospective							
Cr Code 30	05 – Exchange of	Metering Information	- Retrospective				
Cr Code 30 Jurisdiction	05 – Exchange of Objection Logging Business Days	Metering Information Objection Clearing Business Days	- Retrospective Retrospective Business Days	Prospective Business Days			

9.6 Objection Rules

(a) There are no objections.

9.7 Change Request Status Notification Rules

(a) The valid change request status notification rules for the change reason codes identified in clause 9.1 are specified in Table 9-B.

Table 9-B - Change request status notification rules**

Cr Code 3004 – Exchange of Metering Information
Cr Code 3005 – Exchange of Metering Information – Retrospective
PARTICIPANT ROLE & ROLE STATUS – Receives Notification of Change

Status	FRI	ИP	LF	₹	LN	NSP	MD	P	M	РВ	RoL	.R	NSF	2	RP	
Change	N	С	Ν	С	Ν	С	N	С	Ν	С	N	C	N	С	N	С
CANCELLED	-	-	-	-	-	-	-	-	-	Yes	-	-	-	-	-	
COMPLETED	-	Yes	-	Yes	-	Yes	-	Yes	-	Yes	-	-	-	Yes -	•	Yes
OBJECTED	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
PENDING	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
REJECTED	-	-	-	-	-	-	-	-	-	Yes	-	-	-	-	-	-
REQUESTED	-	-	-	-	-	-	-	-			-	-	-	-	-	-

^{**} N = New role, C = Current role.

10. MAINTAIN METERING – CHANGE METERING INSTALLATION DETAILS

10.1 Application [3050 3051]

This procedure applies to the following change reason codes:

• 3050 - Change Metering Installation Details

This is the situation where the current MPB is required to provide a change to the information in the MSATS system in regard to the metering installation details. Each meter serial number must exist in MSATS and a minimum set of metering installation details for the NMI shall exist upon completion of the Change Request. The date at which the information will apply would be a prospective date.

• 3051 - Change Metering Installation Details - Retrospective

This is the situation where the current MPB is required to provide a change to the information in the MSATS system in regard to the metering installation details. Each meter serial number must exist in MSATS and a minimum set of metering installation details for the NMI shall exist upon completion of the Change Request. The date at which the information will apply would be a retrospective date.

10.2 Conditions Precedent

- (a) The NMI exists in the MSATS system.
- (b) The metering installation details do exist in MSATS.
- (c) The NMI classification is wholesale, interconnector, generator or sample.
- (d) There are no objections allowed for this change reason code and NMI classification.
- (e) Chapter 10 relates only to change reason codes 3050 and 3051.

10.3 Initiation Rules

- (a) A current MPB may initiate a change request to change metering installation records in the MSATS system in accordance with clause 10.4
- (b) The current MPB must use one of the following change reason codes 3050 or 3051 to establish an initial change request.
- (c) A minimum set of metering installation details for the NMI shall exist upon completion of the Change Request

10.4 MPB Obligations

The current MPB must:

- (a) Obtain the NMI checksum from an approved source.
- (b) Confirm that the NMI is a valid NMI for the connection point prior to the initiation of a change request.
- (c) Populate an initial change request with the following information:

Change reason code	Participant transaction ID	NMI and NMI checksum
CATS participant ID	Proposed change date	Meter serial ID (at least one)

(d) For all meters associated to the NMI, where the Meter Status Code is to be "C" populate the change request with the following information (where this information does not currently exist in MSATS):

Meter status	Metering installation type code	
--------------	---------------------------------	--

(e) For all Register Identifiers associated to the NMI, where the Register ID Status is to be "C" populate the change request with the following information:

Register ID	Register ID status	Actual / cumulative indicator
Controlled load indicator	Dial format	Multiplier value
Time of day code	Unit of measure code	

The current MPB may:

(f) Populate the initial change request with the following information:

MDM contributory suffix	Network tariff code	Additional site
(for a register)	TOTAL COME	information
Meter location	Meter hazard	Meter route
Meter use	Meter point	Meter manufacturer
Meter model	Transformer location	Transformer type
Transformer ratio	Meter constant	Last test date
Next test date	Test result accuracy	Test result notes
Test performed by	Measurement type	Meter program
Meter read type	Remote phone number	Communication equipment type
Communication protocol	Data conversion arrangements	Data validation arrangements
Estimation instructions	Asset management plan details	Calibration tables (details of any calibration factors programmed into the meter)
Password details (the read and time set passwords only, separated by a space; the write password is not to be recorded in MSATS)	Test and calibration program details	User access rights details (i.e. details of any end- use customer access to the metering installation such as pulse outputs)
Network tariff code additional information	Demand value 1, if the network tariff charge includes a demand component	Demand value 2, if the network tariff charge includes a second demand component

(g) For change reason code 3051, populate the initial change request with:

A atrial and data	
Actual end date	
7 10 10 10 10 10 10 10 10 10 10 10 10 10	

10.5 Time Frame Rules

(a) The valid time frame rules for each jurisdiction for the respective change reason codes are shown in the Table 10-A.

Table 10-A - Time frame rules

CR Code 3050 – Change Meter Details									
Jurisdiction	Objection Logging	Objection Clearing Business	Retrospective	Prospective Business					
	Business Days	Days	Business Days	Days					
ALL	0	0	0	65					
CR Code 3051 - Change a NMI Meter Details - Retrospective									
CR Code 30)51 – Change a N	MI Meter Details - Retro	spective						
CR Code 30 Jurisdiction		MI Meter Details - Retro Objection Clearing Business	<u> </u>	Prospective Business					
			<u> </u>	Prospective Business Days					

10.6 Objection Rules

(a) There are no objections.

10.7 Change Request Status Notification Rules

(a) The valid change request status notification rules for the change reason codes identified in clause 10.1 are specified in Table 10-B.

Table 10-B – Change request status notification rules**

	CR	CR Code 3050 – Change Meter Details CR Code 3051 – Change a NMI Meter Details - Retrospective PARTICIPANT ROLE & ROLE STATUS – Receives Notification of Change														
Status Change	FR	MP	LR		LN	ISP	ME)P	MF	В	Rol	_R	NS	P2	RP	
	N	С	N	С	N	С	N	С	N	С	N	С	N	С	N	С
CANCELLED	-	-	-	-	-	-	-	-	-	Yes	-	-	-	-	-	-
COMPLETED	-	Yes	-	Yes	-	Yes	-	Yes	-	Yes	-	-	-	Yes	-	Yes
OBJECTED	-	-	•	-	ı	-	-	-	-	-	-	-	•	-	•	-
PENDING	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
REJECTED	-	-	-	-	-	-	-	-	-	Yes	-	-	-	-	-	-
REQUESTED	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-

^{*} N = New role, C = Current role.

11. MAINTAIN METERING – CHANGE NETWORK TARIFF CODE

11.1 Application [3100 3101]

This procedure applies to the following change reason codes:

• 3100 - Change network tariff code

This is the situation where the current LNSP is required to provide a change to the information in the MSATS system in regard to the network tariff code details. The date at which the information will apply would be a prospective date.

• 3101 - Change network tariff - retrospective

This is the situation where the current LNSP is required to provide a change to the information in the MSATS system in regard to the network tariff code details. The date at which the information will apply would be a retrospective date.

11.2 Conditions Precedent

- (a) The NMI exists in the MSATS system.
- (b) The NMI classification is wholesale, interconnector, generator or sample.
- (c) There are no objections allowed for this change reason code and NMI classification.
- (d) Chapter 11 relates only to change reason codes 3100 and 3101.

11.3 Initiation Rules

- (a) A current LNSP may initiate a change request to change a network tariff code in accordance with clause 11.4.
- (b) The current LNSP must use one of the following change reason codes 3100 or 3101 to establish an initial change request.

11.4 LNSP Obligations

The current LNSP must:

- (a) Check that the network tariff code created by the MPB when the meter was created in the MSATS system is correct and, if it is not, update it to have the correct value.
- (b) Change the network tariff code in the MSATS system to ensure that the current information provided in the MSATS system is the latest information available from the current LNSP.
- (c) Obtain the NMI checksum from an approved source.
- (d) Confirm that the NMI is a valid NMI for the connection point prior to the initiation of a change request.
- (e) Populate an initial change request with the following information:

Change reason code	Participant transaction ID	NMI and NMI checksum
CATS participant ID	Proposed change date	Meter serial ID
Register ID	Network tariff code	

The current LNSP may:

(f) For change reason codes that are retrospective, populate the initial change request with:

Actual end date		
-----------------	--	--

11.5 Time Frame Rules

(a) The valid time frame rules for each jurisdiction for the respective change reason codes are shown in the Table 11-A.

Table 11-A - Time frame rules

CR Code 3100 – Change network tariff									
Jurisdiction	Objection Logging Business Days	Objection Clearing Business Days	Retrospective Business Days	Prospective Business Days					
ALL	0	0	0	65					
CR Code-3	101 – Change ne	twork tariff – retrospect	tive						
Jurisdiction	Objection Logging	Objection Clearing Business		Prospective Business					
	Business Days	Days	Business Days	Days					
ALL	0	0	140	0					

11.6 Objection Rules

(a) There are no objections.

11.7 Change Request Status Notification Rules

(a) The valid change request status notification rules for the change reason codes identified in clause 11.1 are specified in Table 11-B.

Table 11-B – Change request status notification rules**

	CR C	CR Code 3100 – Change Network Tariff CR Code 3101 – Change Network Tariff - Retrospective PARTICIPANT ROLE & ROLE STATUS – Receives Notification of Change														
Status	FRMF	•	LR		LN	SP	ME)P	MF	В	Ro	LR	NSI	2	RP	
Change	N	С	N	С	N	С	N	С	N	С	N	С	N	С	N	С
CANCELLED	-	-	-	-	-	Yes	-	-	-	-	-	-	-	-	-	-
COMPLETED	-	Yes	-	Yes	-	Yes	-	Yes	-	Yes	-	-	-	Yes	-	Yes
OBJECTED	-	-	-	-	-	-	•	-	-	-	•	-	-	-	-	-
PENDING	-	-	-	-	-	-	•	-	-	-	•	-	-	-	-	-
REJECTED	-	-	-	-	-	Yes	•	-	-	-	•	-	-	-	-	-
REQUESTED	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-

^{**} N = New role, C = Current role.

12. MAINTAIN DATASTREAM – CREATE MDM DATASTREAM

12.1 Application [4000 4001]

This procedure applies to the following change reason codes:

• 4000 - Create MDM Datastream Details

This is the situation where the current MDP establishes the initial set of information in the MSATS system in regard to the MDM datastream details, and there is no associated datastream status change to inactive for the NMI for the same effective date. The date at which the information will apply would be a prospective date.

4001 – Create MDM Datastream Details - Retrospective

This is the situation where the current MDP establishes the initial set of information in the MSATS system in regard to the MDM datastream details, and there is no associated datastream status change to inactive for the NMI for the same effective date. The date at which the information will apply would be a retrospective date.

12.2 Conditions Precedent

- (a) The NMI exists in the MSATS system.
- (b) The MDM Datastream details do not exist in MSATS.
- (c) The NMI classification is wholesale, interconnector, generator or sample.
- (d) There are no objections allowed for this change reason code and NMI classification.
- (e) Chapter 12 relates only to change reason codes 4000 and 4001.

12.3 Initiation Rules

- (a) A current MDP may initiate a change request to create an MDM datastream in the MSATS system in accordance with clause 12.4.
- (b) The current MDP must use one of the following change reason codes 4000 or 4001 to establish an initial change request.

12.4 MDP Obligations

The current MDP must:

- (a) Obtain the NMI checksum from an approved source.
- (b) Confirm that the NMI is a valid NMI for the connection point prior to the initiation of a change request.
- (c) Populate an initial change request with the following information:

Change reason code	Participant transaction ID	NMI and NMI checksum
CATS participant ID	Proposed change date	NMI suffix (at least one)
Datastream type (for each suffix)	Profile name (for each suffix)	Daily average load (for each suffix)
Datastream status code (for each suffix)		

(d) Obtain the average daily load from the LNSP if this value is not otherwise provided by the new FRMP.

The current MDP may:

(e) For change reason codes that are retrospective populate the initial change request with:

Actual end date		
-----------------	--	--

12.5 Time Frame Rules

(a) The valid time frame rules for each jurisdiction for the respective change reason codes are shown in the Table 12-A.

Table 12-A - Time frame rules

CR Code 40	CR Code 4000 – Create MDM Datastream Details								
Jurisdiction	Objection Logging Business Days	Objection Clearing Business Days	Retrospective Business Days	Prospective Business Days					
ALL	0	0	0	65					
CR Code 40	001 - Create MDN	l Datastream Details - R	Retrospective						
Jurisdiction	Objection Leavine	Objection Cleaning Dusiness	Determention						
Julisulction	Objection Logging Business Days	Objection Clearing Business Days	Retrospective Business Days	Prospective Business Days					

12.6 Objection Rules

(a) There are no objections.

12.7 Change Request Status Notification Rules

(a) The valid change request status notification rules for the change reason codes identified in clause 12.1 are specified in Table 12-B.

Table 12-B - Change request status notification rules**

	CR	CR Code 4000 – Create MDM Datastream Details CR Code 4001 – Create MDM Datastream Details - Retrospective PARTICIPANT ROLE & ROLE STATUS – Receives Notification of Change														
Status Change	FR	MP	LR		LN	SP	ME)P	MF	В	Rol	_R	NSP	2	RP	
	N	С	N	С	N	С	N	С	N	С	N	С	N	С	N	С
CANCELLED	-	-	-	-	-	-	-	Yes	-	-	-	-	-	-	-	-
COMPLETED	•	Yes	-	Yes	-	Yes	•	Yes	-	Yes	-	-	-	Yes	-	Yes
OBJECTED	ı	ı	-	-	-	-	•	-	-	-	-	-	-	-	-	-
PENDING	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
REJECTED	ı	ı	-	-	-	-	•	Yes	-	-	-	-	-	-	-	-
REQUESTED	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-

^{**} N = New role, C = Current role.

13. MAINTAIN DATASTREAM - EXCHANGE OF DATASTREAM INFORMATION

13.1 Application [4004 4005]

This procedure applies to the following change reason codes:

• 4004 – Exchange of Datastream Information

This is the situation where the current MDP is required to provide a change to the information in the MSATS system in regard to the MDM datastream details. The change will include at least one datastream status change to inactive and the creation of at least one new datastream. A minimum set of MDM datastream details for the NMI shall exist upon completion of the Change Request. The date at which the information will apply would be a prospective date.

4005 – Exchange of Datastream Information - Retrospective

This is the situation where the current MDP is required to provide a change to the information in the MSATS system in regard to the MDM datastream details. The change will include at least one datastream status change to inactive and the creation of at least one new datastream. A minimum set of MDM datastream details for the NMI shall exist upon completion of the Change Request. The date at which the information will apply would be a retrospective date.

13.2 Conditions Precedent

- (a) The NMI exists in the MSATS system.
- (b) The metering installation details exist in MSATS.
- (c) The NMI classification is wholesale, interconnector, generator or sample.
- (d) There are no objections allowed for this change reason code and NMI classification.
- (e) Chapter 13 relates only to change reason codes 4004 and 4005.

13.3 Initiation Rules

- (a) A current MDP may initiate a change request to exchange datastream records in the MSATS system in accordance with clause 13.4.
- (b) The current MDP must use one of the following change reason codes 4004 or 4005 to establish an initial change request.
- (c) A minimum set of MDM datastream details for the NMI shall exist upon completion of the change request.

13.4 MDP obligations

The current MDP must:

- (a) Obtain the NMI checksum from an approved source.
- (b) Confirm that the NMI is a valid NMI for the connection point prior to the initiation of a change request.
- (c) Populate the change request with the following information:

Change reason code	Participant transaction ID	CATS participant ID
Proposed change date	NMI	NMI checksum

(d) For all datastreams associated to the NMI, where the Datastream Status Code is to be "A" populate the change request with the following information: (where this information does not currently exist in MSATS):

NMI suffix(at least one)	Datastream type (for each suffix)	Profile name (for each suffix)
Daily average load (for each suffix)	Datastream status code (for each suffix)	

The current MDP may:

(e) Populate the change request with the following information:

Meter serial ID (for each	Register ID (for each	MDM contributory suffix
meter)	register ID)	(for each register ID)

(f) For change reason codes that are retrospective, populate the initial change request with:

Actual end date		
-----------------	--	--

13.5 Time Frame Rules

(a) The valid time frame rules for each jurisdiction for the respective change reason codes are shown in the Table 13-A.

Table 13-A - Time frame rules

4004 – Exc	4004 – Exchange of Datastream Information												
Jurisdiction	Objection Logging Business Days	Objection Clearing Business Days	Retrospective Business Days	Prospective Business Days									
ALL	0	0	0	65									
4005 - Exc	4005 – Exchange of Datastream Information – Retrospective												
Jurisdiction	Objection Logging Business Days	Objection Clearing Business Days	Retrospective Business Days	Prospective Business Days									
ALL	0	0	140	0									

13.6 Objection Rules

(a) There are no objections.

13.7 Change Request Status Notification Rules

(a) The valid change request status notification rules for the change reason codes identified in clause 13.1 are specified in Table 13-B.

Table 13-B – Change request status notification rules**

Cr Code 4004 - Exchange of Datastream Information

		Cr Code 4005 – Exchange of Datastream Information - Retrospective															
	PAF	PARTICIPANT ROLE & ROLE STATUS – Receives Notification of Change															
Status	FRI	ΙP	LR		LN	LNSP		MDP		MPB		RoLR		NSP2		RP	
Change	N	С	N	С	Ν	С	N	С	Ν	С	N	С	N	С	N	С	
CANCELLED	-	-	-	-	-	-	-	Yes	-	-	-	-	-	-	-		
COMPLETED	-	Yes	-	Yes	-	Yes	-	Yes	-	Yes	-	-	-	Yes	-	Yes	
OBJECTED	-	-	-	-	-	-	-	-	•	-	-	-	-	-	-	-	
PENDING	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	
REJECTED	-	-	-	-	-	-	-	Yes	-	-	-	-	-	-	-	-	
REQUESTED	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	

^{**} N = New role, C = Current role.

14. MAINTAIN DATASTREAM – CHANGE MDM DATASTREAM

14.1 Application [4050 4051]

This procedure applies to the following change reason codes:

• 4050 - Change MDM Datastream

This is the situation where the current MDP is required to provide a change to the information in the MSATS system in regard to the MDM datastream details. Each Datastream must exist in MSATS and a minimum set of MDM datastream details for the NMI shall exist upon completion of the Change Request. The date at which the information will apply would be a prospective date.

• 4051 - Change MDM Datastream - Retrospective

This is the situation where the current MDP is required to provide a change to the information in the MSATS system in regard to the MDM datastream details. Each Datastream must exist in MSATS and a minimum set of MDM datastream details for the NMI shall exist upon completion of the Change Request. The date at which the information will apply would be a retrospective date.

14.2 Conditions Precedent

- The NMI exists in the MSATS system.
- b) The metering Installation and data stream details exist in MSATS.
- c) The NMI classification is wholesale, interconnector, generator or sample.
- d) There are no objections allowed for this change reason code and NMI classification.
- e) Chapter 14 relates only to change reason codes 4050 and 4051.

14.3 Initiation Rules

- (a) A current MDP may initiate a change request to change an MDM datastream in the MSATS system in accordance with clause 14.4.
- (b) The current MDP must use one of the following change reason codes 4050 or 4051 to establish an initial change request.
- (c) A minimum set of MDM datastream details for the NMI shall exist upon completion of the Change Request.

14.4 MDP Obligations

The current MDP must:

- (a) Obtain the NMI checksum from an approved source.
- (b) Confirm that the NMI is a valid NMI for the connection point prior to the initiation of a change request.
- (c) Populate an initial change request with the following information:

Change reason code	Participant transaction ID	NMI and NMI checksum
CATS participant ID	Proposed change date	NMI suffix (at least one)

(d) For all meters associated to the NMI suffix, where the Datastream Status Code is to be "A", populate the change request with the following information (where this information does not currently exist in MSATS:

Data stream type	Profile name	Daily average load
------------------	--------------	--------------------

Datastream status code

The current MDP may:

(e) Populate the initial change request with the following information:

(f) For change reason codes that are retrospective, populate the initial change request with the following information:

Actual end date	

14.5 Time Frame Rules

(a) The valid time frame rules for each jurisdiction for the respective change reason codes are shown in the Table 14-A.

Table 14-A - Time frame rules

CR Code 4050 – Change MDM Datastream Details												
Jurisdiction	Objection Logging Business Days	Objection Clearing Business Days	Retrospective Business Days	Prospective Business Days								
ALL	0	0	0	65								
CR Code 4051 – Change MDM Datastream Details - Retrospective												
Jurisdiction	Objection Logging Business Days	Objection Clearing Business Days	Retrospective Business Days	Prospective Business Days								
ALL	0	0	140	0								

14.6 Objection Rules

(a) There are no objections.

14.7 Change Request Status Notification Rules

(a) The valid change request status notification rules for the change reason codes identified in clause 14.1 are specified in Table 14-B.

Table 14-B - Change request status notification rules**

CR Code 4050 – Change MDM Datastream Details
CR Code 4051 – Change MDM Datastream Details - Retrospective
PARTICIPANT ROLE & ROLE STATUS – Receives Notification of Change

Status	FRMF	•	LR		LN	SP	ME)P	MF	В	Rol	_R	NSP2	2	RP	
Change	N	С	Ν	С	N	С	N	С	N	С	N	С	N	С	N	С
CANCELLED	-	-	-	-	-	-	-	Yes	-	-	-	-	-	-	-	-
COMPLETED	-	Yes	-	Yes	-	Yes	-	Yes	-	Yes	-	-	-	Yes	-	Yes
OBJECTED	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
PENDING	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
REJECTED	-	-	-	-	-	-	-	Yes	-	-	-	-	-	-	-	-
REQUESTED	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-

^{**} N = New role, C = Current role.

15. MAINTAIN NMI – BACKDATE A NMI

15.1 Application [5001]

This procedure applies to the following change reason code:

5001 – Backdate NMI Start Date

This is the situation where AEMO, on request from an LNSP, or the LNSP establishes the initial set of information in the MSATS system in regard to a connection point. The date at which the information will apply would be a retrospective date.

15.2 Conditions Precedent

- (a) The NMI exists in the MSATS system.
- (b) The NMI classification is wholesale, interconnector, generator or sample.
- (c) Chapter 15 relates only to change reason code 5001.

15.3 Initiation Rules

- (a) AEMO or the LNSP may initiate a change request to backdate a NMI record in the MSATS system in accordance with clause 15.5.
- (b) AEMO or the LNSP must use change reason code 5001 to establish an initial change request.

15.4 LNSP Obligations

The new LNSP (which must be the same party as the current LNSP for the period where the NMI exists in MSATS) must:

- (a) Obtain the NMI checksum, from an approved source.
- (b) Confirm that the NMI is still a valid NMI for the connection point prior to the initiation of a change request.
- (c) Populate an initial change request with values for the following standing data items:

Participant transaction ID	NMI and NMI checksum	CATS participant ID
Proposed change date	Actual end date (which should be the day prior to the day that the existing NMI master record starts on)	TNI code
DLF code	NMI classification code	Jurisdiction code
FRMP	LR	ROLR
RP	MDP	MPB
MPC	LNSP which must be themselves	NMI status code
Locality	State	Postcode

(d) Populate an initial change request with values for the following address fields (as applicable):

EITHER

DPID	Flat number	Flat type
Floor number	Floor type	House number
House number suffix	Location descriptor	Lot number
Street name	Street suffix	Street type

OR

The new LNSP may:

(e) Populate the initial change request with the following information:

Embedded network	Building name	
parent name		

15.5 **AEMO Obligations**

AEMO must:

(a) On request from an LNSP, populate an initial change request with the data provided by the LNSP in accordance with clause 15.4 (CR5001).

15.6 MPB Obligations

The new MPB must:

(a) On receipt of the change request status notification for the completed status, provide the metering installation details to the MSATS system using change reason code 3000 or 3001.

15.7 Time Frame Rules

(a) The valid time frame rules for each jurisdiction for the respective change reason code are shown in the Table 15-A.

Table 15-A - Time frame rules

CR Code 50	CR Code 5001 – Backdate NMI Start Date												
Jurisdiction	Objection Logging	Objection Clearing Business	Retrospective	Prospective Business									
	Business Days	Days	Business Days	Days									
ALL	1	20	140	0									

15.8 Objection Rules

(a) The valid objection rules for the change reason code identified in clause 15.1 are specified in Table 15-B.

Table 15-B - Objection rules**

	CR Code 5	CR Code 5001 – Backdate NMI Start Date														
Objection Code	NMI Class	Jur'n	FRMP		LR		MDF	MDP MPE		3	ROLR		RP		LNSP	
			N	С	N	C	N	C	N	C	N	C	Ν	С	Z	С
NOTRESP	ALL	ALL	Yes	-	Yes	-	Yes	-	Yes	-	Yes	-	-	-	-	-
BADPARTY	ALL	ALL	-	-	-	-	-	-	-	-	-	-	Yes	-		-

^{**} N = New role, C = Current role.

15.9 Change Request Status Notification Rules

(a) The valid change request status notification rules for the change reason code identified in clause 15.1, are specified in Table 15-C.

Table 15-C – Change request status notification rules**

	CR C	CR Code 5001 – Backdate NMI Start Date														
Status	FRMP LR			LNSF	LNSP		MDP		MPB		RoLR		NSP2			
Change	N	С	N	С	N	С	N	С	N	С	N	С	N	С	N	С
CANCELLED	Yes	-	Yes	-	Yes	-	Yes	-	Yes	-	-	-	-	-	Yes	-
COMPLETED	Yes	-	Yes	-	Yes	-	Yes	-	Yes	-	-	-	-	-	Yes	-
OBJECTED	Yes	-	Yes	-	Yes	-	Yes	-	Yes	-	-	-	-	-	Yes	-
PENDING	Yes	-	Yes	-	Yes	-	Yes	-	Yes	-	-	-	-	-	Yes	-
REJECTED	Yes	-	Yes	-	Yes	-	Yes	-	Yes	-	-	-	-	-	Yes	-
REQUESTED	Yes	-	Yes	-	Yes	-	Yes	-	Yes	-	-	-	-	-	Yes	-

^{**} N = New role, C = Current role.

16. MAINTAIN NMI – CHANGE A NMI

16.1 Application [5050 5051]

This procedure applies to the following change reason codes:

5050 – Change NMI Details

This is the situation where the current LNSP is required to provide a change to the set of information in the MSATS system in regard to a connection point. The date at which the information will apply would be a prospective date.

• 5051 - Change NMI Details - Retrospective

This is the situation where the current LNSP is required to provide a change to the information in the MSATS system in regard to a connection point. The date at which the information will apply would be a retrospective date.

16.2 Conditions Precedent

- (a) The NMI exists in the MSATS system.
- (b) The NMI classification is wholesale, interconnector, generator or sample.
- (c) There are no objections allowed for this change reason code and NMI classification.
- (d) Chapter 16 relates only to change reason codes 5050 and 5051.

16.3 Initiation Rules

- (a) A current LNSP may initiate a change request to change a NMI record in the MSATS system in accordance with clause 16.4.
- (b) The current LNSP must use one of the following change reason codes 5050 or 5051 to establish an initial change request.

16.4 LNSP Obligations

The current LNSP must:

- (a) Obtain the NMI checksum from an approved source.
- (b) Confirm that the NMI is a valid NMI for the connection point prior to the initiation of a change request.
- (c) Populate an initial change request with the following information:

Change reason code	Participant transaction ID	NMI and NMI checksum
CATS participant ID	Proposed change date	

The current LNSP may:

(d) Populate the initial change request with the values for following address fields (as applicable):

TNI code	DLF code	NMI classification code
Jurisdiction code	NMI status code	Embedded network parent name
Locality	State	Postcode
Building name	DPID	Flat number
Flat type	Floor number	Floor type
House number	House number suffix	Location descriptor
Lot number	Street name	Street suffix
Street type	Unstructured address 1	Unstructured address 2
Unstructured address 3		

(e) For change reason codes that are retrospective, populate the initial change request with the

Actual end Date		
-----------------	--	--

16.5 Time Frame Rules

(a) The valid time frame rules for each jurisdiction for the respective change reason codes are shown in the Table 16-A.

Table 16-A - Time frame rules

CR Code 5050 - Change NMI Details											
Jurisdiction	Objection Logging	Objection Clearing Business	Retrospective	Prospective Business							
	Business Days	Days	Business Days	Days							
ALL	0	0	0	65							
CR Code 50	51 – Change NM	l Details - Retrospective	е								
Jurisdiction	Objection Logging	Objection Clearing Business		Prospective Business							
	Business Days	Days	Business Days	Days							
ALL	0	0	140	0							

16.6 Objection Rules

(a) The valid objection rules for the change reason codes identified in clause 16.1 are specified in Table 16-B.

Table 16-B - Objection rules**

	CR Code 50	CR Code 5050 - Change NMI Details														
	CR Code 50	CR Code 5051 – Change NMI Details - Retrospective														
Objection Code	NMI Class	Jur'n	FRMP		LR	LR I		OP ME		В	ROLR		RP		LNSP	
Objection Code			N	С	N	С	N	С	N	С	N	С	N	С	N	С
NOTRESP	ALL	ALL	-	-	-	-	-	Yes	-	-	-	-	-	-	-	-
BADMETER	ALL	ALL	-	-	-	-	-	Yes	-	-	-	-	-	-	-	-

^{**} N = New role, C = Current role.
If a cell is blank, no objections are allowed for this role and role status.

16.7 Change Request Status Notification Rules

(a) The valid change request status notification rules for the change reason codes identified in clause 16.1 are specified in Table 16-C.

Table 16-C - Change request status notification rules**

CR Code 5050 – Change NMI Details
CR Code 5051 – Change NMI Details - Retrospective
PARTICIPANT ROLE & ROLE STATUS – Receives Notification of Change

Status Change	FR	MP	LR		LN	SP	ME)P	MF	В	Rol	_R	NS	P2	RP	,
	N	С	N	С	N	С	N	С	N	С	N	С	N	С	N	С
CANCELLED	-	-	-	-	-	Yes	-	-	-	-	-	-	-	-	-	-
COMPLETED	-	Yes	•	Yes	•	Yes	-	Yes	•	Yes	•	-	-	Yes	•	Yes
OBJECTED	•	•	•	•	•	Yes	-	Yes	•	•	ı	-	-	•	•	-
PENDING	•		ı		ı		-		ı	-	ı	-	-		ı	-
REJECTED	•	•	•	•	•	Yes	-	•	•	•	ı	-	-	•	•	-
REQUESTED	-	-	1	-	•	-	-	Yes	•	-	-	-	-	-	•	-

^{**} N = New role, C = Current role.

17. CHANGE ROLE - CHANGE LNSP

17.1 Application [6100 6110]

This procedure applies to the following change reason codes:

6100 – Change LNSP

This is the situation where the new LNSP is required (either by a regulatory change or due to an error in the NMI master record) to provide a change to the information in the MSATS system in regard to the name of the current LNSP. The date at which the information will apply would be a prospective date.

• 6110 - Change LNSP - Retrospective

This is the situation where the new LNSP is required (either by a regulatory change or due to an error in the NMI master record) to provide a change to the information in the MSATS system in regard to the name of the current LNSP. The date at which the information will apply would be a retrospective date.

17.2 Conditions Precedent

- (a) The NMI details exist in MSATS.
- (b) The NMI classification is wholesale, interconnector, generator or sample.
- (c) Chapter 17 relates only to change reason codes 6100 and 6110.

17.3 Initiation Rules

- (a) A new LNSP may initiate a change request to change a LNSP in the MSATS system in accordance with clause 17.4.
- (b) The new LNSP must use one of the following change reason codes 6100 or 6110 to establish an initial change request.

17.4 LNSP Obligations

The new LNSP must:

- (a) Obtain the NMI checksum from an approved source.
- (b) Confirm that the NMI is a valid NMI for the connection point prior to the initiation of a change request.
- (c) Populate an initial change request with the following information:

Change reason code	Participant transaction ID	NMI and NMI checksum
CATS participant ID	Proposed change date	

(d) Nominate itself as the new LNSP.

The new LNSP may:

(e) Populate the initial change request with the following information:

TNI code	DLF code	NMI classification code
Jurisdiction code		

(f) For change reason code that are retrospective, populate the initial change request with:

Actual end date	

The current LNSP may:

(g) Object to the change request in accordance with the objection rules provided in clause 17.6.

17.5 Time Frame Rules

(a) The valid time frame rules for each jurisdiction for the respective change reason codes are shown in the Table 17-A.

Table 17-A - Time frame rules

CR Code 61	CR Code 6100 – Change LNSP								
Jurisdiction	Objection Logging	Objection Clearing Business		Prospective Business					
	Business Days	Days	Business Days	Days					
ALL	1	20	0	65					
CR Code 61	CR Code 6110 – Change LNSP - Retrospective								
Jurisdiction	Objection Logging	Objection Clearing Business	Retrospective	Prospective Business					
	Business Days	Days	Business Days	Days					
ALL	1	20	130	0					

17.6 Objection Rules

(a) The valid objection rules for the change reason codes identified in clause 17.1 are specified in Table 17-B.

Table 17-B - Objection rules**

	CR Code CI	R Code (6100	– CI	hanç	je Li	NSP									
Objection Code	NIMI Class	luwin.	FR	MP	LR		MC)P	MP	В	RO	ROLR			LN	SP
	NMI Class	Jur'n	Ν	С	Ν	O	Z	C	N	С	Ν	O	Ν	O	Ν	C
BADDATA	ALL	ALL	-	-	-	-	•	-	-	-	-	-	-	-	-	Yes
NOTRESP	ALL	ALL	-	-	-	-	-	-	-	-	-	-	-	-	-	Yes
	CR Code 61	10 – Ch	ange	e LN	SP -	Reti	osp	ectiv	/e							
Objection Code	NIMI Class	le color	FR	MP	LR		MC	P	MF	В	RO	LR	RP		LN	SP
Objection Code	NMI Class	Jur'n	Ν	C	Ν	С	N	C	N	С	Ν	C	Ν	C	Ν	С
BADDATA	ALL	ALL	_	_	_	-	-	-	_	-	-	-	-	-	_	Yes
DADDATA	ALL	ALL		_	_											9
NOTRESP	ALL	ALL	-	-	-	-	-	-	-	-	-	-	-	-	-	Yes

^{**} N = New role, C = Current role.

17.7 Change Request Status Notification Rules

(a) The valid change request status notification rules for the change reason codes identified in clause 17.1 are specified in Table 17-C.

Table 17-C - Change request status notification rules**

	CR	CR Code 6100 – Change LNSP CR Code 6110 – Change LNSP - Retrospective PARTICIPANT ROLE & ROLE STATUS – Receives Notification of Change														
Status Change	FR	MP	LR		LNSF	•	MC)P	MF	В	Rol	LR	NS	P2	RP	
	N	С	N	С	N	С	N	С	N	С	N	С	N	С	N	С
CANCELLED	-	-	-	-	Yes	Yes	-	Yes	-	-	-	-	-	-	-	-
COMPLETED	-	Yes	-	Yes	Yes	Yes	-	Yes	-	Yes	-	-	-	Yes	-	Yes
OBJECTED	-	-	-	-	Yes	Yes	-	Yes	-	-	-	-	-	-	-	-
PENDING	-	-	-	-	Yes	Yes	-	Yes	-	-	-	-	-	-	-	-
REJECTED	-	-	-	-	Yes	Yes	-	Yes	-	-	-	-	-	-	-	-
REQUESTED	-	-	-	-	Yes	Yes	-	Yes	-	-	-	-	-	-	-	-

^{**} N = New role, C = Current role.

18. CHANGE ROLE - CHANGE MDP

18.1 Application [6200 6210]

This procedure applies to the following change reason codes:

• 6200 - Change MDP

This is the situation where the current FRMP or current RP is required to provide a change to the information in the MSATS system in regard to the name of the current MDP. The date at which the information will apply would be a prospective date.

• 6210 - Change MDP - Retrospective

This is the situation where the current FRMP or current RP is required to provide a change to the information in the MSATS system in regard to the name of the current MDP. The date at which the information will apply would be a retrospective date.

18.2 Conditions Precedent

- (a) The NMI exists in the MSATS system.
- (b) The NMI classification is wholesale, interconnector, generator or sample.
- (c) Chapter 18 relates only to change reason codes 6200 and 6210.

18.3 Initiation Rules

- (a) The current FRMP or current RP may initiate a change request to change a MDP in the MSATS system in accordance with clause 18.4 or 18.5.
- (b) The current FRMP or current RP must use one of the following change reason codes 6200 or 6210 to establish an initial change request.

18.4 FRMP Obligations

If the change request is initiated by the current FRMP, the current FRMP must:

- (a) Obtain the NMI checksum from an approved source.
- (b) Confirm that the NMI is a valid NMI for the connection point prior to the initiation of a change request.
- (c) Populate an initial change request with the following information:

Change reason code	Participant transaction ID	NMI and NMI checksum
CATS participant ID	Proposed change date	MDP

If the change request is initiated by the current FRMP, the current FRMP may:

(d) For change reason codes that are retrospective, populate the initial change request with the following information:

Actual end date		
-----------------	--	--

18.5 RP Obligations

If the change request is initiated by the current RP, the current RP must:

- (a) Obtain the NMI checksum from an approved source.
- (b) Confirm that the NMI is a valid NMI for the connection point prior to the initiation of a change request.

(c) Populate an initial change request with the following information:

Change reason code	Participant transaction ID	NMI and NMI checksum
CATS participant ID	Proposed change date	MDP

If the change request is initiated by the current RP, the current RP may:

(d) For change reason codes that are retrospective, populate the initial change request with the following information:

Actual end date		
-----------------	--	--

18.6 MDP Obligations

The new and current MDP may:

(a) Object to the change request in accordance with the objection rules provided in clause 18.8.

18.7 Time Frame Rules

(a) The valid time frame rules for each jurisdiction for the respective change reason codes are shown in the Table 18-A.

Table 18-A - Time frame rules

CR Code 62	CR Code 6200 – Change MDP									
Jurisdiction	Objection Logging Business Days	Objection Clearing Business Days	Retrospective Business Days	Prospective Business Days						
ALL	1	20	0	65						
CR Code 62	210 – Change MD	P - Retrospective								
Jurisdiction	Objection Logging	Objection Clearing Business	Retrospective	Prospective Business						
	Business Days	Days	Business Days	Days						
ALL	1	20	130	0						

18.8 Objection Rules

(a) The valid objection rules for the change reason codes identified in clause 18.1 are specified in Table 18-B.

Table 18-B - Objection rules**

	CR Code 6200 - Change MDP															
Objection	NMI	Lunin	FR	MP	LR		MDP		MF	В	RO	LR	RP		LNS	P
Code	Class	Jur'n	Ν	C	N	С	N	С	N	C	Ν	C	Z	С	N	С
BADMETER	ALL	ALL	ı	-		•	Yes	-	ı	-	•	ı	•	•	-	-
DECLINED	ALL	ALL	ı	-	-		Yes	-	ı	-	•	ı	ı		-	-
NOTRESP	ALL	ALL	ı	-	-	•	-	Yes	ı	-	ı	ı	ı	•	-	-
	CR Code 6	210 – C	hang	ge M	DP -	Ret	rospe	ctive								
Objection	NMI	Lunda	FR	FRMP LR I			MDP		MF	В	RO	LR	RP		LNS	P
Code	Class	Jur'n	N	C	N	С	N	С	N	С	N	С	N	С	N	С
BADDATA	ALL	ALL	-	-	-	-	Yes	Yes	-	-	-	-	-	-	-	-
BADMETER	ALL	ALL	-	-	-	-	Yes	-	-	-	-	-		-	-	-
DECLINED	ALL	ALL	-	-	-	-	Yes	-	-	-	-	-	-	-	-	-
NOTRESP	ALL	ALL	-	-	-	-	-	Yes	_	-	-	-	-	-	_	-

^{**} N = New role, C = Current role.

18.9 Change Request Status Notification Rules

(a) The valid change request status notification rules for the change reason codes identified in clause 18.1 are specified in Table 18-C.

Table 18-C - Change request status notification rules**

					•											
	CR	CR Code 6200 – Change MDP														
	CR	CR Code 6210 - Change MDP - Retrospective														
	PA	PARTICIPANT ROLE & ROLE STATUS – Receives Notification of Change														
Status Change	FR	MP	LR		LN	SP	MDF	•	MF	В	RoL	.R	NS	P2	RP	,
	N	С	N	С	N	С	N	С	N	С	N	ပ	Ν	С	N	С
CANCELLED	-	Yes	-	-	-	-	Yes	Yes	-	-	-	-	-	-	-	Yes
COMPLETED	-	Yes	•	Yes	ı	Yes	Yes	Yes	-	Yes	-	-	-	Yes	-	Yes
OBJECTED	-	Yes	-	-	-	-	Yes	Yes	-	-	-	-	-	-	-	Yes
PENDING	-	Yes	-	-	-	-	Yes	Yes	-	-	-	-	-	-	-	Yes
REJECTED	-	Yes	-	-	•	-	-	-	-	-	-	-	-	-	-	Yes
REQUESTED	•	Yes	•	-		-	Yes	Yes	-	-	-	-	-	-	-	Yes

^{**} N = New role, C = Current role.

19. CHANGE ROLE - CHANGE RP

19.1 Application [6300 6301]

This procedure applies to the following change reason codes:

6300 – Change RP

This is the situation where the new RP is required (either by a regulatory change or due to an error in the NMI master record) to provide a change to the information in the MSATS system in regard to name of the current RP. The date at which the information will apply would be a prospective date.

• 6301 - Change RP - Retrospective

This is the situation where the new RP is required (either by a regulatory change or due to an error in the NMI master record) to provide a change to the information in the MSATS system in regard to name of the current RP. The date at which the information will apply would be a retrospective date..

19.2 Conditions Precedent

- (a) The NMI exists in the MSATS system.
- (b) The NMI classification is wholesale, interconnector, generator or sample.
- (c) Chapter 19 relates only to change reason codes 6300 and 6301.

19.3 Initiation Rules

- (a) A new RP may initiate a change request to change the RP in the MSATS system in accordance with clause 19.5.
- (b) The new RP must use one of the following change reason codes 6300 or 6301 to establish an initial change request.

19.4 RP Obligations

The new RP must:

- (a) Obtain the NMI checksum from an approved source.
- (b) Confirm that the NMI is a valid NMI for the connection point prior to the initiation of a change request.
- (c) Populate an initial change request with the following information:

Change reason code	Participant transaction ID	NMI and NMI checksum
CATS participant ID	Proposed change date	

(d) Nominate itself as the new RP.

The new RP may:

(e) For change reason codes that are retrospective, populate the initial change request with the following information:

Actual end date		
-----------------	--	--

The current RP may:

(f) Object to the change request in accordance with the objection rules provided in clause 19.7.

19.5 FRMP Obligations

The current FRMP may:

(a) Object to the change request in accordance with the objection rules provided in clause 19.7.

19.6 Time Frame Rules

(a) The valid time frame rules for each jurisdiction for the respective change reason codes are shown in the Table 19-A.

Table 19-A - Time frame rules

CR Code 6300 – Change RP													
Jurisdiction	Objection Logging	Objection Clearing Business	Retrospective	Prospective Business									
	Business Days	Days	Business Days	Days									
ALL	1	20	0	65									
CR Code 6301 – Change RP - Retrospective													
CR Code 63	801 – Change RP	- Retrospective											
Jurisdiction	Objection Logging	- Retrospective Objection Clearing Business	Retrospective	Prospective Business									
		·	Retrospective Business Days	Prospective Business Days									

19.7 Objection Rules

(a) The valid objection rules for the change reason codes identified in clause 19.2 are specified in Table 19-B.

Table 19-B - Objection rules**

	CR Code 6300 – Change RP															
Objection Code	NMI Class	le cuiles	FR	MP	LR		MD	P	MP	В	RO	LR	RP		LN	SP
Objection Code	NIVII Class	Jur'n	N	С	Ν	С	Ν	С	Ν	С	N	С	N	С	Ν	С
NOTRESP	ALL	ALL	-	-	-	-	-	-	-	-	-	-	-	Yes	-	-
BADPARTY	ALL	ALL	-	Yes	-	-	-	-	-	-	-	-	-	-	-	-
	CR Code 63	01 – Cha	inge	RP - F	Retro	spe	ctive									
Objection Code	NMI Class	le cuiles	FR	MP	LR		MD	P	MP	В	RO	LR	RP		LN	SP
Objection Code	NIVII Class	Jur'n	N	С	Ν	C	N	С	N	С	N	С	N	С	Ν	С
NOTRESP	ALL	ALL	-	-	-	-	-	-	-	-	-	-	-	Yes	-	-
BADPARTY	ALL	ALL	-	Yes	-	-	-	-	-	-	-	-	-	-	-	-
RETRO	ALL	ALL	-	-	-	-	-	-	-	-	-	-	-	Yes	-	-

N = New role, C = Current role.

19.8 Change Request Status Notification Rules

(a) The valid change request status notification rules for the change reason codes identified in clause 19.2 are specified in Table 19-C.

Table 19-C - Change request status notification rules**

	CR Code 6300 – Change RP PARTICIPANT ROLE & ROLE STATUS – Receives Notification of Change															
Status Change	FR	MP	LR		LN	LNSP		MDP		MPB		₹	NSF	2	RP	
	N	С	N	С	N	С	N	С	N	С	N	С	N	С	N	С
CANCELLED	-	Yes	-	-	-	-	-	-	-	-	-	-	-	-	Yes	Yes
COMPLETED	-	Yes	-	Yes	-	Yes	-	Yes	-	Yes	-	-	-	Yes	Yes	Yes
OBJECTED	-	Yes	-	-	-	-	-	-	-	-	-	-	-	-	Yes	Yes
PENDING	-	Yes	-	-	-	-	-	-	-	-	-	-	-	-	Yes	Yes
REJECTED	-	-	-	-	-	-	-	-	-	-	-	-	-	-	Yes	Yes
REQUESTED	-	-	-	-	-	-	-	-	-	-	-	-	-	-	Yes	Yes

CR Code 6301 - Change RP - Retrospective PARTICIPANT ROLE & ROLE STATUS – Receives Notification of Change MPB **FRMP LNSP** MDP **RoLR** NSP2 RP **Status Change** LR N C N С Ν C N C Ν C Ν С N С Ν С CANCELLED Yes Yes Yes COMPLETED Yes Yes Yes Yes Yes Yes Yes Yes Yes OBJECTED Yes PENDING Yes Yes REJECTED Yes Yes **REQUESTED** Yes Yes

N = New role, C = Current role.

20. CHANGE ROLE - CHANGE LR

20.1 Application [6400 6401]

This procedure applies to the following change reason codes:

6400 – Change LR

This is the situation where the new LR is required (either by a regulatory change or due to an error in the NMI master record) to provide a change to the information in the MSATS system in regard to name of the current LR. The date at which the information will apply would be a prospective date.

• 6401 - Change LR - Retrospective

This is the situation where the new LR is required (either by a regulatory change or due to an error in the NMI master record) to provide a change to the information in the MSATS system in regard to name of the current LR. The date at which the information will apply would be a retrospective date.

20.2 Conditions Precedent

- (a) The NMI exists in the MSATS system.
- (b) The NMI classification is wholesale, interconnector, generator or sample.
- (c) Chapter 20 relates only to change reason codes 6400 and 6401.

20.3 Initiation Rules

- (a) A new LR may initiate a change request to change a LR in the MSATS system in accordance with clause 20.4.
- (b) The new LR must use one of the following change reason codes 6400 or 6401 to establish an initial change request.

20.4 LR obligations

The new LR must:

- (a) Obtain the NMI checksum from an approved source.
- (b) Confirm that the NMI is a valid NMI for the connection point prior to the initiation of a change request.
- (c) Populate an initial change request with the following information:

Change reason code	Participant transaction ID	NMI and NMI checksum
CATS participant ID	Proposed change date	

(d) Nominate itself as the new LR.

The new LR may:

(e) For change reason codes that are retrospective, populate the initial change request with the following information:

Actual end date	
Adda ona dato	

The current LR may:

(f) Object to the change request in accordance with the objection rules provided in clause 20.6.

20.5 Time Frame Rules

(a) The valid time frame rules for each jurisdiction for the respective change reason codes are shown in the Table 20-A.

Table 20-A - Time frame rules

CR Code 6400 - Change LR													
Jurisdiction	Objection Logging	Objection Clearing Business	Retrospective	Prospective Business									
	Business Days	Days	Business Days	Days									
ALL	1	20	0	65									
CR Code 6401 – Change LR – Retrospective													
CR Code 64	101 – Change LR	 Retrospective 											
Jurisdiction	Objection Logging	Retrospective Objection Clearing Business	Retrospective	Prospective Business									
			Retrospective Business Days	Prospective Business Days									

20.6 Objection Rules

(a) The valid objection rules for the change reason codes identified in clause 20.1 are specified in Table 20-B.

Table 20-B - Objection rules**

	CR Code 6400 – Change LR															
Objection Code	NMI Class	Jur'n	FRI	MP	LR		MD	P	MP	В	RO	LR	RP		L	SP
Objection Code	NIVII CIASS	Jurn	N	С	N	С	Ν	С	N	С	N	С	N	С	N	С
NOTRESP	ALL	ALL	-	-	-	Yes	ı	-	-	•	-	-	-	ı	•	-
	CR Code 640	01 – Cha	nge l	LR -	Retro	ospec	tive									
Objection Octo	NIMI OLI	la color	FRI	MP	LR		MD	P	MP	В	RO	LR	RP		LN	SP
Objection Code	NMI Class	Jur'n	N	С	N	С	N	С	N	С	N	С	N	С	N	С
NOTRESP	ALL	ALL	-	-	-	Yes	-	-	-	-	-	-	-	-	-	-
RETRO	ALL	ALL	-	-	-	Yes	-	-	-	-	-	-	-	-	-	-

N = New role, C = Current role.

20.7 Change Request Status Notification Rules

The valid change request status notification rules for the change reason codes identified in clause 20.1 are specified in Table 20-C.

Table 20-C - Change request status notification rules**

	CR	Code	6401		ınge	LR - F										
Status Change		ARTICIPANT ROLE & ROLE STATUS – Receives Notification of Change RMP LR LNSP MDP MPB RoLR NSP2 RP														
	N	С	N	С	N	С	N	С	N	С	N	С	N	С	N	С
CANCELLED	-	-	Yes	Yes	-	-	-	Yes	-	-	-	-	-	-	-	-
COMPLETED	-	Yes	Yes	Yes	-	Yes	-	Yes	-	Yes	-	-	-	Yes	-	Yes
OBJECTED	-	-	Yes	Yes	-	-	-	Yes	-	-	-	-	-	-	-	-
PENDING	-	-	Yes	Yes	-	-	-	Yes	-	-	-	-	-	-	-	-
REJECTED	-	-	Yes	Yes	-	-	-	Yes	-	-	-	-	-	-	-	-
REQUESTED	-	-	Yes	Yes	-	-	-	Yes	-	-	-	-	-	-	-	-

^{**} N = New role, C = Current role.

21. CHANGE ROLE – CHANGE LR – EMBEDDED NETWORK (CHILD) – GENERATOR

21.1 Application [6421]

This procedure applies to the following change reason codes:

• 6421 - Change LR - Child NMI - Retrospective

This is the situation where the new LR for an embedded network child NMI is required (due to an error in the NMI master record) to provide a change to the information in the MSATS system in regard to name of the current LR. The date at which the information will apply would be a retrospective date.

21.2 Conditions Precedent

- a) The NMI exists in the MSATS system.
- b) The NMI classification code is GENERATOR.
- c) Chapter 21 relates only to change reason code 6421.
- d) The new LR for the Child must be the FRMP of the Parent NMI.

21.3 Initiation Rules

- A new LR may initiate a change request to change a LR in the MSATS system in accordance with clause 21.4.
- b) The new LR must use the following change reason code 6421 to establish an initial change request.

21.4 LR Obligations

The new LR must:

- a) Obtain the NMI checksum from an approved source.
- Confirm that the NMI is a valid NMI for the connection point prior to the initiation of a change request.
- c) Populate an initial change request with the following information:

Change reason code	Participant transaction ID	NMI and NMI checksum
CATS participant ID	Proposed change date	

d) Nominate itself as the new LR.

The new LR may:

e) Populate the initial change request with the following information:

Actual end date		
-----------------	--	--

The current LR may:

f) Object to the change request in accordance with the objection rules provided in clause 21.6.

21.5 Time Frame Rules

 The valid time frame rules for each jurisdiction for the respective change reason codes are shown in the Table 21-A

Table 21-A - Time frame rules

CR Code 6421 – Change LR Child NMI - Retrospective								
Jurisdiction	Objection Logging Business Days	Objection Clearing Business Days	Retrospective Business Days	Prospective Business Days				
	Generator NMI	Generator NMI						
ALL	1	10	130	0				

21.6 Objection Rules

 The valid objection rules for the change reason codes identified in clause 21.1 are specified in Table 21-B.

Table 21-B - Objection rules**

	CR Cod	CR Code 6421 - Change LR Child NMI - Retrospective														
Objection	NMI	la cultura	FRI	MP	LR		MD	Р	MP	В	MP	С	RP		LNS	SP
Code	Class	Jur'n	N	С	Ν	С	N	С	Ν	С	Ν	С	N	С	N	С
NOTRESP	GEN	ALL	-	-	-	Yes	-	-	-	-	-	-	-	-	-	-
RETRO	GEN	ALL	-	-	-	Yes	-	-	-	•	-	-	•	-	•	-

N = New role, C = Current role.

21.7 Change Request Status Notification Rules

a) The valid change request status notification rules for the change reason codes identified in clause 21.1 are specified in Table 21-C.

Table 21-C- Change request status notification rules**

	CR Code 6421 – Change LR Embedded Network Child – Retrospective PARTICIPANT ROLE & ROLE STATUS – Receives Notification of Change													
Status	FRMF	•	LR		LNS	P	MDP		MPB		MPC		RP	
Change	N	С	N	С	N	С	N	С	N	С	N	С	N	С
CANCELLED	-	-	Yes	Yes	-	-	-	Yes	-	-	-	-	-	-
COMPLETED	-	Yes	Yes	Yes	-	Yes	-	Yes	-	Yes	-	-	-	Yes
OBJECTED	-	-	Yes	Yes	-	-	-	Yes	-	-	-	-	-	-
PENDING	-	-	Yes	Yes	-	-	-	Yes	-	-	-	-	-	-
REJECTED	-	-	Yes	Yes	-	-	-	Yes	-	-	-	-	-	-
REQUESTED	-	-	Yes	Yes	-	-	-	Yes	-	-	-	-	-	-

^{**} N = New role, C = Current role.

22. CHANGE ROLE - CHANGE ROLR

22.1 Application [6500 6501]

This procedure applies to the following change reason codes:

• 6500 - Change ROLR

This is the situation where the new ROLR is required (either by a regulatory change or due to an error in the NMI master record) to provide a change to the information in the MSATS system in regard to name of the current ROLR. The date at which the information will apply would be a prospective date.

6501 – Change ROLR – Retrospective

This is the situation where the new ROLR is required (either by a regulatory change or due to an error in the NMI master record) to provide a change to the information in the MSATS system in regard to name of the current ROLR. The date at which the information will apply would be a retrospective date.

22.2 Conditions Precedent

- (a) The NMI exists in the MSATS system.
- (b) The NMI classification code is wholesale, interconnector, generator or sample.
- (c) Chapter 21 relates only to change reason codes 6500 and 6501.

22.3 Initiation Rules

- (a) A new ROLR may initiate a change request to change a ROLR in the MSATS system in accordance with clause 22.4.
- (b) The new ROLR must use one of the following change reason codes 6500 or 6501 to establish an initial change request.

22.4 ROLR Obligations

The new ROLR must:

- (a) Obtain the NMI checksum from an approved source.
- (b) Confirm that the NMI is a valid NMI for the connection point prior to the initiation of a change request.
- (c) Populate an initial change request with the following information:

Change reason code	Participant transaction ID	NMI and NMI checksum
CATS participant ID	Proposed change date	

(d) Nominate itself as the new ROLR.

The new ROLR may:

(e) For change reason codes that are retrospective, populate the initial change request with the following information:

Actual end date	
Actual Cliu uate	

The current ROLR may:

(f) Object to the change request in accordance with the objection rules provided in clause 22.6.

22.5 Time Frame Rules

(a) The valid time frame rules for each jurisdiction for the respective change reason codes are shown in the Table 22-A.

Table 22-A - Time frame rules

CR Code 6500 – Change ROLR							
Jurisdiction	Objection Logging	Objection Clearing Business	Retrospective	Prospective Business			
	Business Days	Days	Business Days	Days			
ALL	1	20	0	65			
CR Code 6501 – Change ROLR - Retrospective							
CR Code 65	01 – Change ROI	LR - Retrospective					
CR Code 65 Jurisdiction	501 – Change ROI Objection Logging	LR - Retrospective Objection Clearing Business	Retrospective	Prospective Business			
			Retrospective Business Days	Prospective Business Days			

22.6 Objection Rules

(a) The valid objection rules for the change reason codes identified in clause 22.1 are specified in Table 22-B.

Table 22-B - Objection rules**

	CR Code 6500 – Change ROLR															
Objection Code	NMI	Jur'n	FRI	ИP	LR		MDI	0	Μ	РВ	ROI	_R	RP		LNS	SP
Objection Code	Class	Jurn	N	С	N	С	N	С	Ν	С	N	С	N	С	N	С
NOTRESP	ALL	ALL	-	-	-	•	ı	-	•		-	Yes	-	-	-	-
	CR Code	6501 – C	hanç	je RO	LR -	Ret	rospe	ecti	ve							
Objection Ocale	NMI	la color	FRI	ИP	LR		MDI	U	М	РВ	ROI	_R	RP		LNS	SP
Objection Code	Class	Jur'n	N	С	N	С	N	С	Ν	С	N	С	N	С	N	С
NOTRESP	ALL	ALL	-	-	-	-	-	-	-	-	-	Yes	-	-	-	-
RETRO	ALL	ALL	-	-	-	-	1	-	-	•	-	Yes	-	-	-	-

N = New role, C = Current role.

22.7 Change Request Status Notification Rules

(a) The valid change request status notification rules for the change reason codes identified in clause 22.1 are specified in Table 22-C.

Table 22-C - Change request status notification rules**

CR Code 6500 - Change ROLR CR Code 6501 - Change ROLR - Retrospective PARTICIPANT ROLE & ROLE STATUS - Receives Notification of Change NSP2 **RoLR Status Change FRMP** LR **LNSP MDP MPB** RP Ν C Ν C Ν C Ν N C Ν C Ν C Ν C C **CANCELLED** Yes Yes **COMPLETED** Yes Yes Yes Yes Yes Yes Yes Yes Yes OBJECTED Yes Yes PENDING Yes Yes --**REJECTED** ---Yes Yes ---**REQUESTED** -Yes Yes

^{**} N = New role, C = Current role.

23. CHANGE ROLE - CHANGE MPB AND / OR MPC

23.1 Application [6700 6701]

This procedure applies to the following change reason codes:

• 6700 - Change MPB and / or MPC

This is the situation where the current RP is required to provide a change to the information in the MSATS system in regard to name of the current MPB and / or MPC. The date at which the information will apply would be a prospective date.

• 6701 - Change MP - Retrospective

This is the situation where the current RP is required to provide a change to the information in the MSATS system in regard to name of the current MPB and / or MPC. The date at which the information will apply would be a retrospective date.

23.2 Conditions Precedent

- (a) The NMI exists in the MSATS system.
- (b) The NMI classification is wholesale, interconnector, generator or sample.
- (c) Chapter 23 relates only to change reason codes 6700 and 6701.

23.3 Initiation Rules

- (a) A current RP may initiate a change request to change a Metering Provider Category B (MPB) and Metering Data Provider – Category C (MPC) in the MSATS system in accordance with clause 23.4.
- (b) The current RP must use one of the following change reason codes 6700 or 6701 to establish an initial change request.

23.4 RP Obligations

The current RP must:

- (a) Obtain the NMI checksum from an approved source.
- (b) Confirm that the NMI is a valid NMI for the connection point prior to the initiation of a change request.
- (c) Populate an initial change request with the following information:

Change reason code	Participant transaction ID	NMI and NMI checksum
CATS participant ID	Proposed change date	

(d) Populate the initial change request with at least one of the following information:

MPB MPC	
---------	--

The current RP may:

(e) For change reason codes that are retrospective, populate the initial change request with the following information:

	•
Actual end date	
Actual ellu uate	

23.5 MP Obligations

The current MP may:

(a) Object to the change request in accordance with the objection rules provided in clause 23.8.

23.6 LNSP Obligations

The current LNSP may:

(a) Object to the change request in accordance with the objection rules provided in clause 23.8.

23.7 Time Frame Rules

(a) The valid time frame rules for each jurisdiction for the respective change reason codes are shown in the Table 23-A.

Table 23-A - Time frame rules

CR Code 6700 – Change MP										
Jurisdiction	Objection Logging	Objection Clearing Business	pjection Clearing Business Retrospective							
	Business Days	Days	Business Days	Days						
ALL	1	20	0	65						
CR Code 67	CR Code 6701 – Change MP – Retrospective									
Jurisdiction	Objection Logging	Objection Clearing Business	Retrospective	Prospective Business						
o an loaiotion		Objection Cleaning Dusiness	rveirospective	i iospective pusitiess						
Cancalousi	Business Days	Days	Business Days	Days						

23.8 Objection Rules

(a) The valid objection rules for the change reason codes identified in clause 23.1 are specified in Table 23-B.

Table 23-B - Objection rules**

	CR Code 6700 – Change MP CR Code 6701 – Change MP – Retrospective															
Objection Ocale	NMI	l	FRI	/IP	LR		MD	MDP MPB		3	ROLR		RP		LNSP	
Objection Code	Class	Jur'n	N	С	N	С	Ν	С	N	С	N	С	N	С	Z	С
DECLINED	ALL	ALL	-	-	-	-	-	-	Yes	-	-	-	-	-	-	-
NOTAPRD	ALL	ALL	-	-	-	-	-	-	-	-	-	-	-	-	-	Yes

N = New role, C = Current role.

23.9 Change Request Status Notification Rules

(a) The valid change request status notification rules for the change reason codes identified in clause 23.1 are specified in Table 23-C.

Table 23-C – Change request status notification rules**

	CR	Code		– Ch	ange	MP MP – I ROLE S				s Not	ificatio	on of	Chan	ge		
Status	FR	MP	LR		LNS	SP SP	MDP		MPE	3	RoL	R	NSP2		RP	1
Change	N	С	N	С	N	С	N	С	N	С	N	С	N	С	N	С
CANCELLED	-	-	-	-	-	Yes	-	-	Yes	Yes	-	-	-	-	-	Yes
COMPLETED	-	Yes	-	-	-	Yes	-	Yes	Yes	Yes	-	-	-	Yes	-	Yes
OBJECTED	-	-	-	-	-	Yes	-	-	Yes	Yes	-	-	-	-	-	Yes
PENDING	-	-	-	-	-	-	-	-	Yes	Yes	-	-	-	-	-	Yes
REJECTED	-	-	-	-	-	Yes	-	-	Yes	Yes	-	-	-	-	-	Yes
REQUESTED	-	-	-	-	-	Yes	-	-	Yes	Yes	-	-	-	-	-	Yes

^{**} N = New role, C = Current role.

24. AEMO ONLY - AEMO-INITIATED STANDING DATA UPDATES

24.1 Application [5100 5101]

This procedure applies to the following change reason codes:

• 5100 - Correct NMI Master Data - (AEMO only)

This is the situation where AEMO identifies a requirement for a change to a connection point which must either be completed with some urgency (in less time than it would take to change that data item, because of the allowance for an objection period) or where several data items that normally can't be changed in the same transaction must be changed at the same time. The date at which the information will apply would be a prospective date.

• 5101 - Correct NMI Master Data - Retrospective (AEMO only)

This is the situation where AEMO identifies a requirement for a change to a connection point which must either be completed with some urgency (in less time than it would take to change that data item, because of the allowance for an objection period) or where several data items that normally can't be changed in the same transaction must be changed at the same time. The date at which the information will apply would be a retrospective date.

24.2 Conditions Precedent

- (a) The NMI exists in the MSATS system.
- (b) There are no objections allowed for these change reason codes and NMI classification.
- (c) The NMI classification is wholesale, interconnector, generator, eprofile or sample.'
- (d) Chapter 24 relates only to change reason codes 5100 and 5101.

24.3 Initiation Rules

- (a) AEMO may initiate a change request to create a NMI record in the MSATS system in accordance with clause 24.4.
- (b) AEMO must use one of the following change reason codes 5100 or 5101 to establish an initial change request.

24.4 AEMO Obligations

AEMO must:

- (a) Obtain the NMI checksum from an approved source.
- (b) Confirm that the NMI is a valid NMI for the connection point prior to the initiation of a change request.
- (c) Populate an initial change request with the following information:

Change reason code	Participant transaction ID	NMI and NMI checksum
CATS participant ID	Proposed change date	

AEMO may:

(d) Populate the change request with values for the following address fields (as applicable):

TNI code	DLF code	Jurisdiction code
NMI status code	NMI classification code	FRMP
LNSP	LR	RP
МРВ	MDP	MPC
ROLR	Locality	State
Postcode	Building name	
DPID	Flat number	Flat type
Floor number	Floor type	House number
House number suffix	Location descriptor	Lot number
Street name	Street suffix	Street type
Unstructured address 1	Unstructured address 2	Unstructured address 3
Embedded network child name	Embedded network parent name	Aggregate flag

Populate the initial change request with the following information for each datastream:

NMI suffix (at least one)	Datastream type (for each suffix)	Daily average load (for each suffix)
Datastream status code (for each suffix)	Profile name (for each suffix)	

Populate the initial change request with the following information for each meter:

Meter serial ID (at least one)	Metering installation type (for each meter)	Meter status (for each meter)
Additional site information	Next scheduled read date	Meter location
Meter constant	Meter hazard	Meter point
Meter route	Meter use	Meter manufacturer
Meter model	Meter program	Meter read type
Transformer location	Transformer type	Transformer ratio
Measurement type	Last test date	Next test date
Test result accuracy	Test result notes	Test performed by
Remote phone number	Communication equipment type	Communication protocol
Data conversion arrangements	Data validation arrangements	Estimation instructions
Asset management plan details	Calibration tables (details of any calibration factors programmed into the meter)	Password details (the read and time set passwords only, separated by a space; the write password is not to be recorded in MSATS)
Test and calibration program details	User access rights details (i.e. details of any enduse customer access to the metering installation such as pulse outputs)	

Populate the initial change request with the following information for each register:

Register ID (at least one for each meter)	Register ID status (for each register)	Network tariff code
Actual / cumulative indicator (for each register)	Controlled load indicator (for each register)	Network tariff code additional information
MDM contributory suffix (this can also be supplied by the MDP)	Demand value 1, if the network tariff charge includes a demand component	Demand value 2, if the network tariff charge includes a second demand component
Time of day code (for each register)	Dial format (for each register)	Multiplier value (for each register)
Unit of measure code (for each register)		

(e) For change requests that are retrospective, populate the change request with the:

Actual End Date		
-----------------	--	--

24.5 Time Frame Rules

(a) The valid time frame rules for each jurisdiction for the respective change reason codes are shown in the Table 24-A.

Table 24-A - Time frame rules

CR Code 51	CR Code 5100 – Correct NMI Master Data - (AEMO only)									
Jurisdiction	Objection Logging	Objection Clearing Business	Retrospective	Prospective Business						
	Business Days	Days	Business Days	Days						
ALL	0	0	0	65						
CR Code 51	01 - Correct NM	Master Data – Retrosp	ective (AEMO only	y)						
Jurisdiction	Objection Logging	Objection Clearing Business	Retrospective	Prospective Business						
	Business Days	Days	Business Days	Days						
ALL	0	0	150	0						

24.6 Objection Rules

(a) There are no objections.

24.7 Change Request Status Notification Rules

(a) The valid change request status notification rules for the change reason codes identified in clause 24.1 are specified in Table 24-B.

Table 24-B – Change request status notification rules**

	CR Code 5100 – Correct NMI Master Data - (AEMO only) CR Code 5101 – Correct NMI Master Data – Retrospective (AEMO only)															
Status	FRM	IP	LR		LNS	Р	MDF	•	MPE	3	Ro	LR	NSP	2	RP	
Change	N	С	N	С	N	С	N	С	N	С	N	С	N	С	N	С
CANCELLED	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
COMPLETED	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	-	-	Yes	Yes	Yes	Yes
OBJECTED	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
PENDING	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
REJECTED	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
REQUESTED	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-

^{**} N = New role, C = Current role.

25. AEMO ONLY – CHANGE - ROLE, TNI OR DLF – BULK CHANGE TOOL (BCT)

25.1 Introduction

- (a) This procedure covers changes to the following MSATS NMI Standing Data Items
 - LNSF
 - 2. LR
 - 3. FRMP
 - 4. MDP
 - 5. MPB
 - 6. MPC
 - 7. RP
 - 8. ROLR
 - 9. TNI Code
 - 10. DLF Code
- (b) This procedure requires the application of a change reason code. However, the MSATS system provides a specific screen for AEMO as system operator to effect this change of NMI Standing Data under prescribed conditions.
- (c) This is a special transaction that is undertaken by AEMO in accordance with the on-line function provided in MSATS.
- (d) Access rights to the BCT on-line function in MSATS is restricted and will only be available to a person nominated by AEMO.
- (e) When activated, the BCT function will use a unique Change Request Code to update the data specified in the request by AEMO.
- (f) Upon advice from jurisdictions etc of a ROLR event, AEMO will utilise the BCT to manage the associated MSATS Role Changes.
- (g) When the BCT is used in MSATS, active change requests may be changed to cancelled status.
- (h) The BCT ROLR action will change the identity of the current FRMP in the NMI master record to the participant id of the party that the jurisdiction has nominated to undertake the ROLR function.
- (i) Where necessary, the BCT ROLR action will change the identity of the parent LR on the embedded network child NMIs provided these have been set up correctly in MSATS.
- (j) Notification of a change to the NMI master record as per Notifications rules below.
- (k) Responsibilities for requesting, authorising and implementing use of the Bulk Change Tool will be as per Table 25-A.

Table 25-A - Request/Authorise/Implementation Scenarios

Scenario	Request	Authorise	Implement
Mergers & Acquisitions	Participants	AEMO	AEMO
TNI & DLF code updates	Participants	AEMO	AEMO
ROLR Events	AEMO/Jurisdictions	AEMO	AEMO
Any Other Scenario	Jurisdictions/Participants	AEMO	AEMO

25.2 Conditions Precedent

- (a) All Events:
 - 1. The NMI exists in the MSATS system.
 - 2. There are NO objections for the BCT process.
 - 3. The Effective Date of the BCT Change Requests shall always be retrospective.

- 4. The NMI can be any NMI Classification.
- 5. AEMO has agreement on requirements and details from all parties impacted.
- The use of the BCT functionality is not dependant on the provision of any meter readings into MSATS.

(b) ROLR Event:

- A decision on the registration status of a current FRMP will be required from AEMO's Board or an authorised person from a Jurisdiction before AEMO can invoke the ROLR function within the MSATS system.
- (c) Mergers & Acquisitions:
 - The Participant Ids involved in the BCT process must have the same ABN and Jurisdictional licences.

25.3 AEMO Obligations

AEMO must:

- (a) Ensure all affected participants have provided approval for change in cases where the BCT is to be used as a result of a participant request.
- (b) Ensure all affected participants are aware of the proposed use of the BCT and its criteria where initiated by a Jurisdiction.
- (c) Ensure BCT data complies with the criteria set by the participants or jurisdictions requesting the change.

AEMO may:

(d) Initiate a BCT change, in the case of a ROLR event, based on direction from the AEMO Board.

25.4 Requesting Participant Obligations

The Requesting Participant must:

- (a) Provide at least one data item (data field) for change.
- (b) Specify the Start and End Date (if required) for the BCT process.
- (c) Ensure that all changing data items are active for the whole of the time period specified for the BCT.
- (d) Ensure all impacted participants are consulted and have provided approval for change to AEMO.
- (e) Specify the notifications to be sent out for the BCT process.
- (f) Specify the action required to be taken when a conflicting Change Request is found for the NMI's in the BCT process.
- (g) Provide the selection criteria for NMIs to be included in the BCT from the following fields:

LNSP	LR	FRMP
MDP	MPB	MPC
RP	ROLR	TNI Code
DLF Code	Post Code	Aggregate Flag
NMI Status Code	NMI Class Code	Jurisdiction Code
Embedded Network Parent Code (Embedded network ID or "if null" or "if not null")	Embedded Network Child Code (Embedded network ID or "if null" or "if not null")	The Tier Status (not required if both the LR and FRMP are provided as selection criteria)
NMI Limit (Maximum number of NMIs to be processed in the run)	Previous Bulk Change run identifier (i.e. Change all NMIs in a previous BCT event)	

(h) Nominate at least one of the following information for change via the BCT:

LNSP	LR	FRMP
MDP ⁽¹⁾	MPB ⁽¹⁾	MPC ⁽¹⁾
RP	ROLR	TNI Code
DLF Code		

Note(1): The Requesting Participant must be the RP to nominate the MDP, MPB and MPC roles.

The Requesting Participant may:

(i) Further specify a list of NMIs to be used as part of the selection criterion.

25.5 Impacted Participant/s Obligations

The Impacted Participant/s must:

- (a) Provide AEMO acceptance or rejection of the criteria set by the Initiating Participant for use with the BCT (other than a ROLR event).
- (b) Where rejecting the criteria set by the Initiating Participant for use with the BCT, provide AEMO reasonable justification for the rejection.

25.6 MDP Obligations

The MDP must:

 (a) Update the activity status of any affected datastreams via normal Change Request processes.

25.7 Time Frame Rules

- (a) Change Requests initiated via the BCT can only be retrospective.
- (b) All Retrospective Days shall apply as per the appropriate Change Reason Code that would normally have applied (refer to appropriate change request Chapter in this document).
- (c) No objection logging or clearing periods shall be available.

25.8 Objection Rules

(a) There are no objections.

25.9 Change Request Status Notification Rules

(a) The Change Request status notification rules for BCT processes are configurable to suit the changes being made. In each case these will need to be specified by the initiating participant.

26. WIGS PROCEDURES VERSION HISTORY

Version	Comments
1.0	Final Determination released on 18 March 2004.
	Version 1.0 is effective from 27 July 2004.
	The effective version 1.0 also contains minimal textual changes to
	accommodate the extension of retail competition in Queensland and the provision of Tasmania in the MSATS system. These Queensland and
	Tasmanian changes were included after a formal Code consultation.
1.1	Final Determination released on 4 February 2005.
1.1	Version 1.1 is effective from 30 March 2005.
	Version 1.1 contains changes to accommodate the provision of Tasmania in
	the MSATS system.
1.11 Update to reflect	The WIGS 1.11 procedures incorporate revised wording as a consequence of
National Electricity Rules	the change from the National Electricity Code to National Electricity Rules e.g.
23 September 2005	"Code" replaced by "Rules", "Code Participants" replaced by "Registered
	Participants", and "NECA" replaced by "AEMC". The National Electricity Rules
1.2 Final Determination	commenced on 1 July 2005.
26 September 2005	This version of the WIGS Procedures is released to include further process and improvement suggestions following operational experience.
1.3 Final Determination	This version of the WIGS Procedures is released to include further process
19th May 2006	and improvement suggestions following operational experience.
2.7 Final Determination	This version of the WIGS Procedures is released to include further process
1st July 2007	and improvement suggestions following operational experience and to
3	facilitate the introduction of FRC in Queensland.
2.8 Final Determination	This version of the WIGS Procedures is released to include further process
28th May 2008	and improvement suggestions following operational experience.
2.9 Final Determination	This version of the WIGS Procedures is released to include further process
5th September 2008	and improvement suggestions following operational experience.
2.10 Final Determination	This version of the WIGS Procedures is released to include further process
6th January 2009	and improvement suggestions following operational experience.
2.10.1 Final Determination	This version of the WIGS Procedures is released to include further process
14th May 2009 3.0 Final Determination	and improvement suggestions following operational experience. This version of the WIGS Procedures is released to include further process
5th June 2009	and improvement suggestions following operational experience.
3.1 Final Determination	This version of the WIGS Procedures is released to include further process
7th August 2009	and improvement suggestions following operational experience and the
7.1.7.tagaot 2000	introduction of Advanced Metering.
3.2 Final Determination	This version of the WIGS Procedures is released to include further process
8 th April 2010	and improvement suggestions following operational experience.

3.3 Final Determination	This version of the WIGS Procedures is released to include further process
25th October 2010	and improvement suggestions following operational experience.
3.4 Final Determination	This version of the WIGS Procedures is released to include further process
21st December 2010	and improvement suggestions following operational experience and the
	introduction of Advanced Metering.
3.4a	This version of the WIGS Procedures is released to facilitate further extension
15 th July 2011	of contestability to small business customers in Tasmania.
3.5 Final Determination	This version of the WIGS Procedures is released to include a new Change
15 th August 2011	Request CR6421 and manifest changes.
3.6 Final Determination	This version of the WIGS Procedures is released to include manifest changes
13th January 2012	to Change Request 6421 to reflect that the CR is limited to Generators only.
3.7 Final Determination	This version of the WIGS Procedures is released to include a new NMI
15 th May 2013	Discovery Type 3 transaction; aligning procedures with the NER as a result of
	introduction of new terminology regarding MDPs; updates based on a New
	Connections process review to facilitate that a LNSP only nominate a retailer
	as the FRMP based on instruction from that retailer; changes as a result of
	introduction of Market Small Generation Aggregator (MSGA).
3.8 Final Determination	This version of the WIGS Procedures is released to align MPB obligations with
13 th November 2013	MDP obligations during the meter set up or maintenance process, by requiring
	the MPB to source the suffix from the appropriate MDP.
3.9 Final Determination	No version 3.9 was released. The Tasmanian Government withdrew its
1 st January 2014	request for the change to the WIGS Procedures.
4.0 Final Determination	This version of the WIGS Procedures is released to include amendments
15 th May 2014	arising from the introduction of NECF in NSW.
4.1 Final Determination	This version of the WIGS Procedures is released to include amendments
1 July 2014	arising from the AEMC Rule change on Access to NMI Standing Data.

DISCLAIMER

This document is made available to you on the following terms and conditions. If you do not agree to the terms and conditions please notify AEMO on telephone 1300 858 724 and immediately return this document to AEMO, GPO Box 2008, Melbourne, VIC 3001.

- (a) **Purpose -** This document is provided to you for information purposes only. You are not permitted to commercialise this document or the information contained in it.
- (b) **Reliance** This document may be subsequently amended and therefore your reliance on the contents of this document is at your own risk.
- (c) Confidential Information This document and the information contained in it is the confidential information of AEMO. You must not disclose the confidential information to any person without the prior written consent of AEMO unless the confidential information is legally required to be disclosed.
 - The person or body who has been permitted access to this document by AEMO may disclose this document to its employees, contractors and advisers provided the document is only made available on these terms and conditions and this disclaimer is attached to the document or any part of it which is disclosed.
- (d) Intellectual Property AEMO Limited is the owner of the copyright in this document. All rights reserved. All material is subject to copyright under the Copyright Act 1968 (Commonwealth) and permission must be obtained in writing from AEMO Limited. AEMO is the owner of all other intellectual property rights in this document and the information contained in it. You must not in any way or by any means store, reproduce or modify such contents without AEMO's express written consent.
- (e) No Warranty Neither AEMO nor any of AEMO's advisers, consultants or other contributors to this document (or their respective associated companies, businesses, partners, directors, officers or employees) make any representation or warranty, express or implied, as to the currency, accuracy, reliability or completeness of the information.
- (f) No Liability To the maximum extent permitted by law, neither AEMO nor any of its advisers, consultants or other contributors to this document (or their respective associated companies, businesses, partners, directors, officers or employees) shall have any liability (whether arising from negligence or otherwise) in respect of the use by you of the information (including any reliance on the currency, accuracy, reliability or completeness of the information).

Confidential Information of AEMO.

© 2014- AEMO Limited is the owner of the copyright in this document. All rights reserved. All material is subject to copyright under the Copyright Act 1968 (Commonwealth) and permission must be obtained in writing from AEMO Limited.